#### Free standard shipping on all orders







Search by keywords, subject, or ISBN



#### ADD TO WISH LIST

ISBN 9780367618810 434 Pages 11 Color & 91 B/W Illustrations Published January 29, 2024 by CRC Press

#### **<u>Request Inspection Copy</u>**

Free Shipping (14-21 Business Days) <u>shipping options</u>

### Description

The book provides insights in the decision-making for implementing strategies in various spheres of real-world issues. It integrates optimal policies in various decisionmaking problems and serves as a reference for researchers and industrial practitioners. Furthermore, the book provides sound knowledge of modelling of real-world problems and solution procedure using the various optimisation and statistical techniques for making optimal decisions.

The book is meant for teachers, students, researchers and industrialists who are working in the field of materials science, especially operations research and applied statistics.

### Table of Contents

1. A New Version of the Generalized Rayleigh Distribution with Copula, Properties, Applications and Different Methods of Estimation

M Masoom Ali, Haitham M Yousof and Mohamed Ibrahim

2. Expanding the Burr X Model: Properties, Copula, Real Data Modeling and Different Methods of Estimation

M Masoom Ali, Mohamed Ibrahim and Haitham M Yousof

3. Transmuted Burr Type X Model with Applications to Life Time Data

Tabassum Naz Sindhu, Zawar Huassian and Muhammad Aslam

4. Monitoring Patients Blood Level through Enhanced Control Chart

Muhammad Aslam, Khushnoor Khan and Nasrullah Khan

5. Goodness of Fit in Parametric and Non-parametric Econometric Models

Shalabh, Subhra Sankar Dhar and N Balakrishna

6. Stochastic Models for Cancer Progression and its Optimal Programming for Control with Chemotherapy

Tirupathi Rao Padi

7. A New Unrelated Question Model with Two Questions Per Card

Tonghui Xu, Stephen A Sedory and Sarjinder Singh

8. Hybrid of Simple Model and a New Unrelated Question Model for Two Sensitive Characteristics

Renhua Zheng, Stephen A Sedory and Sarjinder Singh

9. Hybrid of Crossed Model and a New Unrelated Question Model for Two Sensitive Characteristics

Renhua Zheng, Stephen A Sedory and Sarjinder Singh

10. Modified Regression Type Estimator by Ingeniously Utilizing Probabilities for more Efficient Results in Randomized Response Sampling

Roberto Arias, Stephen A Sedory and Sarjinder Singh

11. Ratio and Regression Type Estimators for a New Measure of Coefficient of Dispersion Relative to the Empirical Mode

Christin Variathu Eappen, Stephen A Sedory and Sarjinder Singh

12. Class of Exponential Ratio Type Estimator for Population Mean in Adaptive Cluster Sampling

Akingbade Toluwalase Janet and Balogun Oluwafemi Samson

13. An Inventory Model for Substitutable Deteriorating Products under Fuzzy and Cloud Fuzzy Demand Rate

Nita H Shah and Milan B Patel

14. Co-ordinated Selling Price and Replenishment Policies for Duopoly Retailers under Quadratic Demand and Deteriorating

#### Serial No. 416 Nature of Items

Nita H Shah and Monika K Naik

**15.** Quadratic Programming Approach for the Optimal Multiobjective Transportation Problem

Masar Al-Rabeeah, Ali Al-Hasani and M G M Khan

16. Analyzing Multi-Objective Fixed-Charge Solid Transportation Problem under Rough and Fuzzy-Rough Environments

Sudipta Midya and Sankar Kumar Roy

17. Overall Shale Gas Water Management: A Neutrosophic Optimization Approach

Ahmad Yusuf Adhami, Firoz Ahmad and Nahida Wani

18. Memory Effect on an EOQ Model with Price Dependant Demand and Deterioration

Mostafijur Rahaman, Sankar Prasad Mondal and Shariful Alam

19. Optimality Conditions of an Unconstrained Imprecise Optimization Problem via Interval Order Relation

Md Sadikur Rahman and Asoke Kumar Bhunia

20. Power Comparison of Different Goodness of Fit Tests for Beta Generalized Weibull Distribution

Kanchan Jain, Neetu Singla and Suresh K Sharma

21. On the Transmuted Modified Lindley Distribution: Theory and Applications to Lifetime Data

Lishamol Tomy, Christophe Chesneau and Jiju Gillariose

22. Adjusted Bias and Risk for Estimating Treatment Effect after Selection with an Application in Idiopathic Osteoporosis

Omer Abdalghani, Mohd Arshad, K R Meena and A K Pathak

23. Validity Judgement of an EOQ Model using Phi-coefficient

*Suman Maity, Sujit Kumar De, Madhumangal Pal and Sankar Prasad Mondal* 

24. Uncertain Chance-Constrained Multi-Objective Geometric Programming Problem

Sahidul Islam

25. Optimal Decision Making for the Prediction of Diabetic Retinopathy in Type 2 Diabetes Mellitus Patients

*Faiz Noor Khan Yusufi, Nausheen Hashmi, Aquil Ahmed and Jamal Ahmad* 

### Editor(s)

#### Biography

**Irfan Ali** received his B.Sc., M.Sc., M.Phil., and Ph.D. degrees from Aligarh Muslim University, Aligarh. He is currently working as an Assistant Professor with the Department of Statistics and Operations Research, A.M.U., Aligarh. His research interests include applied statistics and optimization. He has published more than 80 research articles in reputed journals and serves as a reviewer for several journals. He has delivered invited lectures in several universities and Institutions. He was awarded with Post Graduate Merit Scholarship during his M.Sc. (statistics) and the UGC-BSR Scholarship during his Ph.D. (statistics).

**Leopoldo Eduardo Cardenas-Barron** is a researcher of the research group in Optimization and Data Science. He is a member of the Mexican Research National System since 2015. Earlier, in 1996, had joined the Industrial and Systems Engineering Department at Tecnológico de Monterrey, where he was appointed to Assistant Professor in 1997, Associate Professor in 2001, and Professor in 2007.

**Aquil Ahmed** obtained his Ph.D. from the University of Roorkee (Presently 1.1.T., Roorkee). He is the Founder Head, Department of Statistics and Former Dean, Faculty of Material Science, University of Kashmir, Srinagar. He served the University of Kashmir, Srinagar for more than 26 years before joining the Department of Statistics and Operations Research, Aligarh Muslim University, Aligarh in 2014. He has also served at the Asian Institute of Technology, Bangkok, Thailand and Qassim University, K.S.A. He is currently working as the Chairman, Department of Statistics and Operations Research, A.M.U., Aligarh.

**Ali Akbar Shaikh** is Assistant Professor in Mathematics at The University of Burdwan, Burdwan, West Bengal, India. He was a postdoctoral fellow at School of Engineering and Sciences of Tecnológico de Monterrey, Mexico. He was awarded SNI of level 1 (out of 0-3) by National System of Researchers of Mexico, the government of Mexico in the year 2017.

#### Free standard shipping on all orders







Search by keywords, subject, or ISBN



#### ADD TO WISH LIST

ISBN 9780367618810 434 Pages 11 Color & 91 B/W Illustrations Published January 29, 2024 by CRC Press

#### **<u>Request Inspection Copy</u>**

Free Shipping (14-21 Business Days) <u>shipping options</u>

### Description

The book provides insights in the decision-making for implementing strategies in various spheres of real-world issues. It integrates optimal policies in various decisionmaking problems and serves as a reference for researchers and industrial practitioners. Furthermore, the book provides sound knowledge of modelling of real-world problems and solution procedure using the various optimisation and statistical techniques for making optimal decisions.

The book is meant for teachers, students, researchers and industrialists who are working in the field of materials science, especially operations research and applied statistics.

### Table of Contents

1. A New Version of the Generalized Rayleigh Distribution with Copula, Properties, Applications and Different Methods of Estimation

M Masoom Ali, Haitham M Yousof and Mohamed Ibrahim

2. Expanding the Burr X Model: Properties, Copula, Real Data Modeling and Different Methods of Estimation

M Masoom Ali, Mohamed Ibrahim and Haitham M Yousof

3. Transmuted Burr Type X Model with Applications to Life Time Data

Tabassum Naz Sindhu, Zawar Huassian and Muhammad Aslam

4. Monitoring Patients Blood Level through Enhanced Control Chart

Muhammad Aslam, Khushnoor Khan and Nasrullah Khan

5. Goodness of Fit in Parametric and Non-parametric Econometric Models

Shalabh, Subhra Sankar Dhar and N Balakrishna

6. Stochastic Models for Cancer Progression and its Optimal Programming for Control with Chemotherapy

Tirupathi Rao Padi

7. A New Unrelated Question Model with Two Questions Per Card

Tonghui Xu, Stephen A Sedory and Sarjinder Singh

8. Hybrid of Simple Model and a New Unrelated Question Model for Two Sensitive Characteristics

Renhua Zheng, Stephen A Sedory and Sarjinder Singh

9. Hybrid of Crossed Model and a New Unrelated Question Model for Two Sensitive Characteristics

Renhua Zheng, Stephen A Sedory and Sarjinder Singh

10. Modified Regression Type Estimator by Ingeniously Utilizing Probabilities for more Efficient Results in Randomized Response Sampling

Roberto Arias, Stephen A Sedory and Sarjinder Singh

11. Ratio and Regression Type Estimators for a New Measure of Coefficient of Dispersion Relative to the Empirical Mode

Christin Variathu Eappen, Stephen A Sedory and Sarjinder Singh

12. Class of Exponential Ratio Type Estimator for Population Mean in Adaptive Cluster Sampling

Akingbade Toluwalase Janet and Balogun Oluwafemi Samson

13. An Inventory Model for Substitutable Deteriorating Products under Fuzzy and Cloud Fuzzy Demand Rate

Nita H Shah and Milan B Patel

14. Co-ordinated Selling Price and Replenishment Policies for Duopoly Retailers under Quadratic Demand and Deteriorating

#### Serial No. 416 Nature of Items

Nita H Shah and Monika K Naik

**15.** Quadratic Programming Approach for the Optimal Multiobjective Transportation Problem

Masar Al-Rabeeah, Ali Al-Hasani and M G M Khan

16. Analyzing Multi-Objective Fixed-Charge Solid Transportation Problem under Rough and Fuzzy-Rough Environments

Sudipta Midya and Sankar Kumar Roy

17. Overall Shale Gas Water Management: A Neutrosophic Optimization Approach

Ahmad Yusuf Adhami, Firoz Ahmad and Nahida Wani

18. Memory Effect on an EOQ Model with Price Dependant Demand and Deterioration

Mostafijur Rahaman, Sankar Prasad Mondal and Shariful Alam

19. Optimality Conditions of an Unconstrained Imprecise Optimization Problem via Interval Order Relation

Md Sadikur Rahman and Asoke Kumar Bhunia

20. Power Comparison of Different Goodness of Fit Tests for Beta Generalized Weibull Distribution

Kanchan Jain, Neetu Singla and Suresh K Sharma

21. On the Transmuted Modified Lindley Distribution: Theory and Applications to Lifetime Data

Lishamol Tomy, Christophe Chesneau and Jiju Gillariose

22. Adjusted Bias and Risk for Estimating Treatment Effect after Selection with an Application in Idiopathic Osteoporosis

Omer Abdalghani, Mohd Arshad, K R Meena and A K Pathak

23. Validity Judgement of an EOQ Model using Phi-coefficient

*Suman Maity, Sujit Kumar De, Madhumangal Pal and Sankar Prasad Mondal* 

24. Uncertain Chance-Constrained Multi-Objective Geometric Programming Problem

Sahidul Islam

25. Optimal Decision Making for the Prediction of Diabetic Retinopathy in Type 2 Diabetes Mellitus Patients

*Faiz Noor Khan Yusufi, Nausheen Hashmi, Aquil Ahmed and Jamal Ahmad* 

### Editor(s)

#### Biography

**Irfan Ali** received his B.Sc., M.Sc., M.Phil., and Ph.D. degrees from Aligarh Muslim University, Aligarh. He is currently working as an Assistant Professor with the Department of Statistics and Operations Research, A.M.U., Aligarh. His research interests include applied statistics and optimization. He has published more than 80 research articles in reputed journals and serves as a reviewer for several journals. He has delivered invited lectures in several universities and Institutions. He was awarded with Post Graduate Merit Scholarship during his M.Sc. (statistics) and the UGC-BSR Scholarship during his Ph.D. (statistics).

**Leopoldo Eduardo Cardenas-Barron** is a researcher of the research group in Optimization and Data Science. He is a member of the Mexican Research National System since 2015. Earlier, in 1996, had joined the Industrial and Systems Engineering Department at Tecnológico de Monterrey, where he was appointed to Assistant Professor in 1997, Associate Professor in 2001, and Professor in 2007.

**Aquil Ahmed** obtained his Ph.D. from the University of Roorkee (Presently 1.1.T., Roorkee). He is the Founder Head, Department of Statistics and Former Dean, Faculty of Material Science, University of Kashmir, Srinagar. He served the University of Kashmir, Srinagar for more than 26 years before joining the Department of Statistics and Operations Research, Aligarh Muslim University, Aligarh in 2014. He has also served at the Asian Institute of Technology, Bangkok, Thailand and Qassim University, K.S.A. He is currently working as the Chairman, Department of Statistics and Operations Research, A.M.U., Aligarh.

**Ali Akbar Shaikh** is Assistant Professor in Mathematics at The University of Burdwan, Burdwan, West Bengal, India. He was a postdoctoral fellow at School of Engineering and Sciences of Tecnológico de Monterrey, Mexico. He was awarded SNI of level 1 (out of 0-3) by National System of Researchers of Mexico, the government of Mexico in the year 2017.

### Serial No. 417 **SPRINGER NATURE** Link

| $\equiv$ Menu Q Search  | 다. Cart |
|---|---------|
| Home > Book   |         |
| France: Character Later spectra<br>A data water spectra<br>A data water spectra<br>A data water between the spectra<br>B spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectra<br>Spectr |         |

### **Computational Modelling in Industry 4.0**

### A Sustainable Resource Management Perspective

Book © 2022

### **Overview**

Editors: Irfan Ali, Prasenjit Chatterjee, Ali Akbar Shaikh, Neha Gupta, Ali AlArjani

Contains latest research contributions on sustainable resource management combined with Industry 4.0

Addresses the different problems, practices, challenges and opportunities in

sustainable resource management

Includes the case study of manufacturing and other related industries

Solitations ← 1 Altmetric



This is a preview of subscription content, log in via an institution [2] to check

access.

#### Serial No. 417 Department of Statistics and Operations Research, Aligarh Muslim University, Aligarh, India

Irfan Ali

#### Department of Mechanical Engineering, MCKV Institute of Engineering, West Bengal, India

Prasenjit Chatterjee

### Department of Mathematics, University of Burdwan, West Bengal, India

Ali Akbar Shaikh

Amity School of Business, Amity University, Noida, India

Neha Gupta

#### Industrial Engineering Department, Prince Sattam Bin Abdulaziz University, Alkharj, Saudi Arabia

Ali AlArjani

### **About the editors**

Dr. Irfan Ali received B.Sc., M.Sc., M.Phil., and Ph.D. degrees from Aligarh Muslim University. He is currently working as an Assistant Professor with the Department of Statistics and Operations Research, Aligarh Muslim University. He received the Post Graduate Merit Scholarship Award during M.Sc. (statistics) and the UGC-BSR Scholarship awarded during the Ph.D. (statistics) programs in 2013. His research interests include applied statistics, survey sampling, supply chain networks and management, mathematical programming, and multiobjective optimization. He has supervised M.Sc., M.Phil., and Ph.D. students in operations research. He has completed a research project UGC– Start-Up Grant Project, UGC, New Delhi, India. He has published more than 75 research articles in reputed journals and serves as a Reviewer for several journals. He is currently editing two books to be published by Taylor France and Springer Nature. He is a Lifetime Member of various professional societies: Operational Research Society of India, Indian Society for Probability and

Statistics, Indian Mathematical Society, and The Indian Science Congress Association. He delivered invited talks in several universities and institutions. He also serves as an Associate Editor for some journals.

Dr. Prasenjit Chatterjee is currently the Dean (Research and Consultancy) at MCKV Institute of Engineering, West Bengal, India. He has over 100 research papers in various international journals and peer-reviewed conferences including International Journal of Production Research, International Journal of Intelligent Systems, Expert Systems with Applications, Operations Management Research, Applied Soft Computing, Computers and Industrial Engineering, Socio-Economic Planning Sciences, Management Decision, Clean Technologies and Environmental Policy, Journal of Cleaner Production, Journal of Natural Fibers, Benchmarking: an International Journal, OPSEARCH, International Journal of Advanced Manufacturing Technology, Materials and Design, Robotics and Computer Integrated Manufacturing to name a few. He has authored and edited more than 15 books on intelligent decision-making, supply chain management, optimization techniques, risk and sustainability modelling. He has received numerous awards including Best Track Paper Award, Outstanding Reviewer Award, Best Paper Award, Outstanding Researcher Award and University Gold Medal. Dr. Chatterjee is the Editor-in-Chief of Journal of Decision Analytics and Intelligent Computing. He has also been the Guest Editor of several special issues in different SCIE / Scopus / ESCI (Clarivate Analytics) indexed journals. He is the Lead Series Editor of "Disruptive Technologies and Digital Transformations for Society 5.0", Springer. He is also the Lead Series Editor of "Concise Introductions to AI and Data Science", Scrivener - Wiley; AAP Research Notes on Optimization and Decision Making Theories; Frontiers of Mechanical and Industrial Engineering, Apple Academic Press, co-published with CRC Press, Taylor and Francis Group and "River Publishers Series in Industrial Manufacturing and Systems Engineering". Dr. Chatterjee is one of the developers of two multiple-criteria decision-making methods called Measurement of Alternatives and Ranking according to COmpromise Solution (MARCOS) and Ranking of Alternatives

through Functional mapping of criterion sub-intervals into a Single Interval (RAFSI).

**Dr. Ali Akbar Shaikh** is an Assistant Professor of Mathematics at The University of Burdwan, West Bengal, India. Earlier, he was a Postdoctoral Fellow at the School of Engineering and Sciences of Tecnológico de Monterrey, México. He has obtained the award SNI of level 1 (out of 0−3) presented by the National System of Researchers of México from Government of México in 2017. He obtained his PhD and MPhil in Mathematics from The University of Burdwan, and MSc in Applied Mathematics from University of Kalyani, India. He has published more than 55 research papers in different international journals of repute. His research interests include inventory control, interval optimisation, and particle swarm optimisation.

Dr. Neha Gupta is presently working as an Assistant Professor at Amity School of Business, AUUP. She obtained her M.Sc. (Operations Research), M.Phil. and PhD (Operations Research) from Aligarh Muslim University, Aligarh (INDIA). Her broad research area includes optimization and decision sciences. She is an Editorial Member of International Journal of Mathematics and Systems Science and International Journal of Data Mining, Modelling & Management, Inderscience Publications. Recently She has edited a special issue for Journal of Revenue and Pricing Management, Springer, and authored two books for international publishers like Taylor and Francis. She is a life member of the Operational Research Society of India and has more than 30 publications in journals of national and international repute. She has participated in several conferences of national and international level.

**Dr. Ali AlArjani** is an Assistant Professor at the Industrial Engineering Department at Prince Sattam bin Abdulaziz University, Alkharj, KSA. He did a Bachelor of Science degree in Mechanical Engineering from King Fahad University for Petroleum and Minerals, and master and PhD in Industrial Engineering from the USA. Dr. AlArjani's research background is in industrial

optimization, mathematical modelling, energy, data clustering analysis and machine learning classification and prediction applications.

## **Bibliographic Information**

**Book Title** Computational Modelling in Industry 4.0 **Book Subtitle** A Sustainable Resource Management Perspective

**DOI** https://doi.org/10.1007/9 78-981-16-7723-6 **Publisher** Springer Singapore

Hardcover ISBN

**Edition Number** 

2022

1

978-981-16-7722-9

Published: 13 February

**Copyright Information** The Editor(s) (if applicable) and The Author(s), under exclusive license to Springer Nature Singapore Pte Ltd. 2022

eBook ISBN 978-981-16-7723-6 Published: 12 February 2022

Number of Illustrations 24 b/w illustrations, 86 illustrations in colour Topics Industrial and Production Engineering, Industrial Chemistry/Chemical Engineering, Materials Science, general **Editors** Irfan Ali, Prasenjit Chatterjee, Ali Akbar Shaikh, Neha Gupta, Ali AlArjani

eBook Packages Engineering, Engineering (R0)

Softcover ISBN 978-981-16-7725-0 Published: 14 February 2023

Number of Pages XVI, 366



Advances in Heat Transfer Volume 58, 2024, Pages 81-154

## Chapter Three - Advances in nanofluid flow, heat, and mass transfer at moving/stretching surfaces

Kuppalapalle Vajravelu <sup>a</sup> ♀ ⊠, John Patrick Abraham <sup>b</sup>, Swati Mukhopadhyay <sup>c</sup>, P. Lakshminarayana <sup>d</sup>

- <sup>a</sup> Department of Mathematics, Department of Mechanical, Material & Aerospace Engineering, University of Central Florida, Orlando, FL, United States
- <sup>b</sup> Department of Mechanical Engineering, School of Engineering, University of St. Thomas, St Paul, MN, United States
- <sup>c</sup> Department of Mathematics, The University of Burdwan, Burdwan, West Bengal, India
- <sup>d</sup> Department of Mathematics, Vellore Institute of Technology, Vellore, Tamil Nadu, India

Available online 25 June 2024, Version of Record 26 September 2024.

? What do these dates mean?

Show less 🔨

😪 Share 🍠 Cite

https://doi.org/10.1016/bs.aiht.2024.05.003 ⊅ Get rights and content ⊅

### Abstract

This book chapter is both a review of the current state-of-the-art and a report on future areas of research. It focuses on nanofluid flow, heat, and mass transfer due to moving/stretching surfaces. This chapter contains five sections; the first section provides details about the nanofluids, hybrid nanofluids, and advancements in their utilization. In the second section, non-Newtonian nanofluid flow past an unsteady stretching sheet is analyzed in detail along with the consequences of an externally applied magnetic field,

K Biotechnology and Biological Sciences (https://www.taylorfrancis.com/books/mono/10.1201/9781003001614/biotechnology-biological-sciences?refid=b8644e71-def0-4b75-a254-d1d49190a1cb&context=ubx)

Show Path 🗸

Chapter



#### Statistical design in accelerated downstream processing

By Arshad Jawed (/search?contributorName=Arshad Jawed&contributorRole=author&redirectFromPDP=true&context=ubx), Mohd Wahid (/search? contributorName=Mohd Wahid&contributorRole=author&redirectFromPDP=true&context=ubx), Payel Chaudhury (/search?contributorName=Payel Chaudhury&contributorRole=author&redirectFromPDP=true&context=ubx), Shafiul Haque (/search?contributorName=Shafiul Haque&contributorRole=author&redirectFromPDP=true&context=ubx), Sajad A.Dar (/search?contributorName=Sajad A.Dar&contributorRole=author&redirectFromPDP=true&context=ubx), Shikha Joon (/search?contributorName=Shikha Joon&contributorRole=author&redirectFromPDP=true&context=ubx)

Book <u>Biotechnology and Biological Sciences (https://www.taylorfrancis.com/books/mono/10.1201/9781003001614/biotechnology-biological-sciences?</u> refld=9fab4128-f16d-4c49-85a7-347a435d2b8e&context=ubx)

| 1st Edition   |
|---------------|
| 2019          |
| CRC Press     |
| 12            |
| 9781003001614 |
|               |

😪 Share

#### ABSTRACT

< Previous Chapter (chapters/edit/10.1201/9781003001614-31/exploration-molecular-docking-dynamic-simulation-studies-identify-potential-cid-50906864-potential-anti-angiogenic-molecule-faizan-ahmad-lakshmi?context=ubx)

Next Chapter > (chapters/edit/10.1201/9781003001614-33/study-antioxidant-properties-bioconverted-agricultural-waste-debarati-roy-sukanya-chakraborty? context=ubx)



## Serial No. 420 SPRINGER NATURE Link

 $\equiv$  Menu

**Q** Search

ঢ় Cart

Home > Oxidative Stress in Microbial Diseases > Chapter

### Oxidative Stress in Entamoeba histolytica

| Chapter | First Online: 27 October 2019

pp 257–280 Cite this chapter

SetHaden Gert Neter Deservice Street Oxidative Stress in Microbial Diseases **Oxidative Stress in Microbial** 

<u>Diseases</u>

Somasri Dam 🖂, Pinaki Biswas & Raktim Ghosh

**547** Accesses **4** <u>Citations</u>

### Abstract

*Entamoeba histolytica* is a human pathogen, responsible for invasive amoebiasis and dysentery. This chapter aims to describe the effect of various stresses especially oxidative and nitrosative stress on this organism. This parasite is subjected to several types of stress throughout its life cycle and also during the invasion of human tissues as a result of host's response to the infection. For successful infection, it must produce an adaptive response against host defense mechanisms. *E. histolytica* is microaerophilic, but during tissue invasion, it is exposed to high oxygen content in well-perfused tissues. This parasite has its own antioxidant strategy to protect itself against reactive oxygen and nitrogen species generated by both host and parasite. *E. histolytica* doesn't have most of the antioxidant defense mechanisms such as glutathione peroxidase, glutathione reductase, and catalase. Instead, it manages the antioxidant components from engulfed

and EhNO. Later on, H<sub>2</sub>O<sub>2</sub> produces H<sub>2</sub>O by EhPrx which is reduced by Eh34 or Trx. Fd can act as an alternative source of electron to facilitate the reduction of O<sub>2</sub> to H<sub>2</sub>O without ROS generation via FDP system. FDP enzyme system can also detoxify the NO-generated by the parasite or from host immune response. Reduced Trx is converted to its oxidized form by TrxR through NADPH oxidation. EhSOD works to clean the harmful intracellular O<sub>2</sub><sup>-</sup> by ORP. EhRbr with association of NROR protect mitosomes by converting H<sub>2</sub>O<sub>2</sub> into H<sub>2</sub>O. ISF and HCP have potential antioxidant capacity. *E. histolytica* can also decrease the redox potential of extracellular environment through the trans-PMET.

### **Author information**

### **Authors and Affiliations**

Department of Microbiology, University of Burdwan, Burdwan, West Bengal, India Somasri Dam, Pinaki Biswas & Raktim Ghosh

### **Corresponding author**

Correspondence to Somasri Dam.

### **Editor information**

### **Editors and Affiliations**

Department of Biochemistry and Biophysics, University of Kalyani, Kalyani, West Bengal, India Sajal Chakraborti

Department of Biochemistry and Biophysics, University of Kalyani, Kalyani, West Bengal, India Tapati Chakraborti

**Amity University, Kolkata, West Bengal, India** Dhrubajyoti Chattopadhyay

National Institute of Immunology, New Delhi, India Chandrima Shaha

### **Rights and permissions**

### Serial No. 420 Reprints and permissions

Copyright information

© 2019 Springer Nature Singapore Pte Ltd.

### About this chapter

### Cite this chapter

Dam, S., Biswas, P., Ghosh, R. (2019). Oxidative Stress in *Entamoeba histolytica*. In: Chakraborti, S., Chakraborti, T., Chattopadhyay, D., Shaha, C. (eds) Oxidative Stress in Microbial Diseases. Springer, Singapore. https://doi.org/10.1007/978-981-13-8763-0\_14

.RIS坐 .ENW坐 .BIB坐

DOI https://doi.org/10.1007/9 78-981-13-8763-0\_14

Published 27 October 2019 Publisher Name Springer, Singapore

Print ISBN 978-981-13-8762-3

Online ISBN 978-981-13-8763-0 eBook Packages <u>Biomedical and Life</u> <u>Sciences</u> <u>Biomedical and Life</u> <u>Sciences (R0)</u>

### Publish with us

Policies and ethics [7

## Serial No. 421 SPRINGER NATURE Link

 $\equiv$  Menu

**Q** Search

Home > Handbook of Oxidative Stress in Cancer: Therapeutic Aspects > Reference work entry

### Microbe-Induced Oxidative Stress in Cancer Development and Efficacy of Probiotics as Therapeutics in Preventing Its Onset and Progression

| Reference work entry | First Online: 29 September 2022

| pp 3513–3542 | <u>Cite this reference work entry</u>

Handbook of Oxidative Stress in

**Cancer: Therapeutic Aspects** 

#### Pinaki Biswas, Suchetana Pal, Moubonny Das & Somasri Dam 🖂

**95** Accesses

### Abstract

Within the metaorganisms, cross-communication between commensal organisms and the host is crucial for the maintenance of physiological homeostasis. At times, microbiota becomes accountable for breaching homeostasis by creating a microenvironment favoring uncontrolled cell growth. Chronic infection and inflammation act as inciting factors for reactive oxygen species generation promoting damage to DNA, proteins, and lipids. Oxidative stress activates a variety of transcription factors that in turn control the expression of several inflammatory cytokines and chemokines. Consecutive changes



Safe Ondorson False Handbook of Oxidative Stress in Cancer: Therapeutic Aspects

The financial assistance from DAE-BRNS, Government of India (Project No. 37 (1)/20/15/2014-BRNS), and CSIR, Government of India (Project No. 27(0322)/17/EMR-II) is thankfully acknowledged. **Author information** 

### **Authors and Affiliations**

Department of Microbiology, The University of Burdwan, Burdwan, India Pinaki Biswas, Suchetana Pal, Moubonny Das & Somasri Dam

### **Corresponding author**

Correspondence to Somasri Dam.

### **Editor information**

### **Editors and Affiliations**

Department of Biochemistry and Biophysics, University of Kalyani, Kalyani, West Bengal, India Sajal Chakraborti

### **Section Editor information**

Chair and Department of Toxicology, Jagiellonian University Collegium Medicum, Kraków, Poland Maria Walczak Department of Biochemistry and Biophysics, University of Kalyani, Kalyani, West Bengal, India

Tapati Chakraborti

| Department of Physiology, Anatomy and Genetics, University of Oxford, Ox | ford, UK |
|--|----------|
| Pulak Kar  |          |

Department of Microbiology, The University of Burdwan, Bardhaman, West Bengal, India Somasri Dam

Institute of Veterinary Physiology, University of Zurich, Zurich, Switzerland

Kuntal Dey

Internal Medicine, The University of Texas MD Anderson Cancer Center, Houston, USA Muthusamy Kunnimalaiyaan

### **Rights and permissions**

**Reprints and permissions** 

### **Copyright information**

© 2022 Springer Nature Singapore Pte Ltd.

### About this entry

### Cite this entry

Biswas, P., Pal, S., Das, M., Dam, S. (2022). Microbe–Induced Oxidative Stress in Cancer Development and Efficacy of Probiotics as Therapeutics in Preventing Its Onset and Progression. In: Chakraborti, S. (eds) Handbook of Oxidative Stress in Cancer: Therapeutic Aspects. Springer, Singapore. https://doi.org/10.1007/978-981-16-5422-0\_159

.RIS ± .ENW ± .BIB ±

| DOI<br>https://doi.org/10.1007/9<br>78-981-16-5422-0_159 | Published<br>29 September 2022   | Publisher Name<br>Springer, Singapore  |
|--|----------------------------------|--|
| Print ISBN<br>978-981-16-5421-3                          | Online ISBN<br>978-981-16-5422-0 | eBook Packages<br><u>Biomedical and Life</u><br><u>Sciences</u><br><u>Reference Module</u><br><u>Biomedical and Life</u><br>Sciences |





In Health and Disease

## Advances in Health and Disease. Volume 54

\$250.00 - \$375.00

Lowell T. Duncan

Series: Advances in Health and Disease BISAC: MED000000 DOI: https://doi.org/10.52305/TZFI6457

This book presents some recent updates in health and disease. Chapter One provides an overview of facioscapulohumeral muscular dystrophy. Chapter Two summarizes the recent research findings related to the pathogenesis of E. histolytica. Chapter Three summarizes the literature concerning the role of fluoride in osteogenesis. Chapter Four investigates the utility of ultrasound elastography in detection of liver disease. Chapter Five evaluates the efficiency of health services of some countries during the COVID-19 pandemic. Chapter Six describes treatment options for petroclival meningiomas. Chapter Seven explains the uses, mechanisms of actions and clinical outcomes of ibuprofen. Chapter Eight defines and describes limb-girdle muscular dystrophies, including its clinical manifestations, diagnostic strategies, and treatment options. Finally, Chapter Nine discusses the diagnosis of methamphetamine-related poisoning and death, including pathophysiological and toxicological findings, in forensic practices.

\*\*Order both the printed version and the e-book together and SAVE!\*\*

Binding Hardcover ✔ € Clear

Status: AV

Pages: 234

\$250.00

| 1 | Add to | o cart |
|---|--------|--------|
|   |        |        |

🔿 Add to Wishlist

ISBN: 978-1-68507-868-3

Categories: 2022, Advances in Health and Disease (Numbered Series), Books, Medicine and Health, Nova Medicine and Health, Special Topics

Like 0



### Table of Contents

#### Preface

Chapter 1. An Overview of Facioscapulohumeral Muscular Dystrophy

Raymond A. Huml<sup>1</sup>, MS, DVM, RAC, Teresa Leon<sup>2</sup>, MD, PhD, and Maryna Kolochavina<sup>3</sup>, PharmD, PMP

<sup>1</sup>Vice President of Medical and Scientific Strategy, Head of the Rare Disease Consortium, Syneos Health<sup>®</sup> Clinical Solutions Morrisville, North Carolina, USA

 $\wedge$ 

<sup>2</sup>Madrid, Spain

<sup>3</sup>Berlin, Germany

Chapter 2. Understanding *Entamoeba histolytica* Pathogenesis: In Light of Recent Advancements

Suchetana Pal, Raktim Ghosh and Somasri Dam Department of Microbiology, The University of Burdwan, Burdwan, West Bengal, India **Chapter DOI:** https://doi.org/10.52305/BOVY9060

Amanda Souza Scotti<sup>1</sup>, Melissa Camassola<sup>1</sup>, Juliana da Silva<sup>2,3</sup> and Ana Letícia Hilario Garcia<sup>2,3</sup> <sup>1</sup>Laboratory of Stem Cells and Tissue Engineering PPGBioSaúde (Postgraduate Program in Cellular and Molecular Biology Applied to Health), Lutheran University of Brazil (ULBRA), Canoas, Rio Grande do Sul, Brazil

<sup>2</sup>Laboratory of Genetic Toxicology, PPGBioSaúde, Lutheran University of Brazil (ULBRA), Canoas, RS, Brazil

<sup>3</sup>Laboratory of Genetic Toxicology, La Salle University (UniLaSalle), Canoas, RS, Brazil

#### Chapter 4. Utility of Ultrasound Elastography in Detection of Liver Disease in Areca Nut Habitual

Suwarna Dangore- Khasbage, MDS, PhD, MPhil (HPE)

Fellowship in Head & Neck Imaging, Professor & Head, Dept. of Oral Medicine & Radiology, Sharad Pawar Dental College & Hospital, Datta Meghe Institute of Medical Sciences (Deemed to be University), Sawangi (Meghe) Wardha, India

#### Chapter 5. Efficiency Evaluation of Health Services of Selected Countries during the COVID-19 Pandemic

N. Firat Ozkan and Berna H. Ulutas

Department of Industrial Engineering, Eskisehir Osmangazi University, Eskisehir, Turkey

#### Chapter 6. Petroclival Meningiomas

Tyler Warner<sup>1</sup>, MD, Modhi Alhussinan<sup>2</sup>, MD, Sami Khairy<sup>3</sup>, MD, Faisal Farrash<sup>4</sup>, MD, and Turki Elarjani1, MD

<sup>1</sup>Department of Neurosurgery, University of Miami, Miami, Fl, USA

<sup>2</sup>Department of Neurosurgery, King Fahad Medical City, Riyadh, Saudi Arabia

<sup>3</sup>Department of Neurosurgery, King Abdulaziz Medical City, National Guard, Riyadh, Saudi Arabia

<sup>4</sup>Department of Neurosurgery, King Faisal Specialist Hospital and Research Centre, Riyadh, Saudi Arabia

#### Chapter 7. Ibuprofen: Uses, Mechanisms of Action and Clinical Outcomes

Gustavo Henrique de Mattos-Pereira<sup>1</sup>, Rafael Paschoal Esteves Lima<sup>1</sup> and Fernando Oliveira Costa<sup>1</sup>, PhD

<sup>1</sup>Department of Clinical Dentistry, Pathology and Oral Surgery, Faculty of Dentistry, Federal University of Minas Gerais, Belo Horizonte, Brazil

へ

#### Chapter 8. Limb-Girdle Muscular Dystrophy: A Review

Nermina Nakas

Executive Director, Neuroscience and Rare Disease, Medical and Scientific Management, Clinical Solutions, Syneos Health<sup>®</sup>, Morrisville, North Carolina USA

## Serial No. 423 SPRINGER NATURE Link

| <u></u> ■ Menu   | Q Search | 🕂 Cart |
|--|----------|--------|
| Home > Tex   | xtbook   |        |
| Advanced Operations<br>Advanced Operations<br>Research |          |        |
| 🕘 Springer   |          |        |

### Advanced Optimization and Operations Research

| Textbook | © 2019

### Overview

Authors: Asoke Kumar Bhunia, Laxminarayan Sahoo, Ali Akbar Shaikh

Serves as a textbook for optimization and operations research Discusses topics like dynamic and fuzzy inventory models, infinite game, flow network and optimization of queuing problems Includes a number of solved problems and detailed examples in all chapters Is suitable for coursework and independent study Is fashioned for easy comprehension of the subject matter

Part of the book series: <u>Springer Optimization and Its Applications</u> (SOIA, volume 153)

Login



This is a preview of subscription content, log in via an institution [2] to check

access.





**Buy Hardcover Book** 

Tax calculation will be finalised at checkout

Other ways to access

Licence this eBook for your library  $\rightarrow$ 

Institutional subscriptions  $\rightarrow$ 

### About this book

This textbook provides students with fundamentals and advanced concepts in optimization and operations research. It gives an overview of the historical perspective of operations research and explains its principal characteristics, tools, and applications. The wide range of topics covered includes convex and concave functions, simplex methods, post optimality analysis of linear programming problems, constrained and unconstrained optimization, game theory, queueing theory, and related topics. The text also elaborates on project management, including the importance of critical path analysis, PERT and CPM techniques. This textbook is ideal for any discipline with one or more courses in optimization and operations research; it may also provide a solid reference for researchers and practitioners in operations research.

### Similar content being viewed by others







| Serial No. 423          | Introduction to | Mathematical Decision- |
|-------------------------|-----------------|------------------------|
| of linear programming:  | Optimization    | Convex Programming     |
| surprising applications |                 | convex Programming     |
| Article Open access     | Chapter © 2019  | Chapter © 2017         |
| 21 September 2024       |                 |                        |

### Keywords



### Table of contents (18 chapters)

**Front Matter** 



#### Introduction to Operations Research

Asoke Kumar Bhunia, Laxminarayan Sahoo, Ali Akbar Shaikh Pages 1-11

#### **Convex and Concave Functions**

Asoke Kumar Bhunia, Laxminarayan Sahoo, Ali Akbar Shaikh Pages 13-27

#### **Simplex Method**

Asoke Kumar Bhunia, Laxminarayan Sahoo, Ali Akbar Shaikh Pages 29-78

#### **Revised Simplex Method**

Asoke Kumar Bhunia, Laxminarayan Sahoo, Ali Akbar Shaikh Pages 79-108

#### **Dual Simplex Method**

Asoke Kumar Bhunia, Laxminarayan Sahoo, Ali Akbar Shaikh Pages 109-125

#### **Bounded Variable Technique**

Asoke Kumar Bhunia, Laxminarayan Sahoo, Ali Akbar Shaikh Pages 127-136

#### Post-optimality Analysis in LPPs

Asoke Kumar Bhunia, Laxminarayan Sahoo, Ali Akbar Shaikh



#### **Integer Programming**

Asoke Kumar Bhunia, Laxminarayan Sahoo, Ali Akbar Shaikh Pages 171-195

#### **Basics of Unconstrained Optimization**

Asoke Kumar Bhunia, Laxminarayan Sahoo, Ali Akbar Shaikh Pages 197-210

#### **Constrained Optimization with Equality Constraints**

Asoke Kumar Bhunia, Laxminarayan Sahoo, Ali Akbar Shaikh Pages 211-247

#### **Constrained Optimization with Inequality Constraints**

Asoke Kumar Bhunia, Laxminarayan Sahoo, Ali Akbar Shaikh Pages 249-267

#### **Quadratic Programming**

Asoke Kumar Bhunia, Laxminarayan Sahoo, Ali Akbar Shaikh Pages 269-302

#### **Game Theory**

Asoke Kumar Bhunia, Laxminarayan Sahoo, Ali Akbar Shaikh Pages 303-366

#### **Project Management**

Asoke Kumar Bhunia, Laxminarayan Sahoo, Ali Akbar Shaikh



#### **Queueing Theory**

Asoke Kumar Bhunia, Laxminarayan Sahoo, Ali Akbar Shaikh Pages 403-486

#### **Flow in Networks**

Asoke Kumar Bhunia, Laxminarayan Sahoo, Ali Akbar Shaikh Pages 487-519

#### **Inventory Control Theory**

Asoke Kumar Bhunia, Laxminarayan Sahoo, Ali Akbar Shaikh Pages 521-579

#### **Mathematical Preliminaries**

Asoke Kumar Bhunia, Laxminarayan Sahoo, Ali Akbar Shaikh Pages 581-613

**Back Matter** 

Download chapter PDF 坐

Pages 615-621

Back to top 1

### **Authors and Affiliations**

#### Department of Mathematics, The University of Burdwan, Rajbati, India

Asoke Kumar Bhunia, Ali Akbar Shaikh

Department of Mathematics, Raniganj Girls' College, Searsole Rajbari, India

### Serial No. 423 Laxminarayan Sahoo

### About the authors

ASOKE KUMAR BHUNIA is Professor at the Department of Mathematics, The University of Burdwan, West Bengal, India. He obtained his Ph.D. degree in Mathematics and M.Sc. in Applied Mathematics from Vidyasagar University, India. His research interests include computational optimization, soft computing, interval mathematics, and interval ranking. He has published over 125 research papers in various national and international journals of repute. He is a reviewer of several SCI journals. He has guided 13 Ph.D. and two M.Phil. students. An author of four research monographs and six book chapters, he is an INSA visiting fellow and former Associate Editor of the Springer's journal, *OPSEARCH*.

LAXMINARAYAN SAHOO is Assistant Professor at the Department of Mathematics, Raniganj Girls' College, West Bengal, India. He obtained his Ph.D. in Mathematics from The University of Burdwan and M.Sc. in Applied Mathematics from Vidyasagar University, India. His research interests include reliability optimization, computational optimization, artificial intelligence, soft computing, interval mathematics, interval ranking, and fuzzy mathematics. He has published several research papers in various national and international journals of repute. He has received MHRD fellowship from the Government of India and the Prof. M.N. Gopalan Award for the Best Ph.D. Thesis in Operations Research from the Operational Research Society of India.

ALI AKBAR SHAIKH is Assistant Professor of Mathematics at The University of Burdwan, West Bengal, India. Earlier, he was a postdoctoral fellow at the School of Engineering and Sciences of Tecnológico de Monterrey, México. He has obtained the award SNI of level 1 (out of 0–3) presented by the National System of Researchers of México from the Government of México in the year 2017. He earned his Ph.D. and M.Phil. degrees in Mathematics from The University of

Burdwan, and M.Sc. in Applied Mathematics from the University of Kalyani, India. Dr. Shaikh has published over 37 research papers in different international journals of repute. His research interests include inventory control, interval optimization, and particle swarm optimization.

## **Bibliographic Information**

**Book Title** Advanced Optimization and Operations Research **Authors** Asoke Kumar Bhunia, Laxminarayan Sahoo, Ali Akbar Shaikh

DOI https://doi.org/10.1007/9 78-981-32-9967-2

**Copyright Information** Springer Nature Singapore Pte Ltd. 2019 Hardcover ISBN 978-981-32-9966-5 Published: 30 January 2020 Series Title Springer Optimization and Its Applications

eBook Packages Mathematics and Statistics, Mathematics and Statistics (R0)

**Softcover ISBN** 978-981-32-9969-6 Published: 26 August 2021

eBook ISBN 978-981-32-9967-2 Published: 09 January 2020

**Edition Number** 

1

Series ISSN 1931-6828

Publisher

Springer Singapore

Series E-ISSN 1931-6836

Number of Pages XVI, 621 Number of Illustrations 88 b/w illustrations

| Topics                     |
|----------------------------|
| Operations Research,       |
| Management Science,        |
| Calculus of Variations and |



Phospholipases in Physiology and Pathology Volume 7, 2023, Pages 9-33

# Chapter 2 - Pathological aspects of microbial phospholipases

Raktim Ghosh \*, Suchetana Pal \*, Subhasish Sarkar, Somasri Dam

Department of Microbiology, The University of Burdwan, Burdwan, West Bengal, India

Available online 18 August 2023, Version of Record 18 August 2023.

? What do these dates mean?

Show less 🔨

i≡ Outline 🛛 🖧 Share 🗦 Cite

https://doi.org/10.1016/B978-0-443-21800-2.00017-8 ↗ Get rights and content ↗

#### Abstract

<u>Phospholipases</u> are a ubiquitous group of <u>enzymes</u> that break down <u>phospholipids</u> into fatty acids and other compounds, and play crucial roles in several <u>biochemical processes</u> and intracellular <u>signaling pathways</u>. Microbial phospholipases are implicated to be associated with the pathogenesis of numerous diseases. These enzymes may act directly on the host cell membrane and cause cell damage by membrane destruction or by interfering with the host's internal signal transduction systems. This chapter describes the pathophysiological roles of phospholipases in the pathogenesis of various diseases caused by bacteria, fungi, and <u>protozoa</u>. Understanding the biological role of microbial phospholipases and the development of inhibitors for these enzymes are absolutely necessary to generate potential therapeutics and vaccines. This will ultimately reduce the consequences of associated diseases in both humans and animals.
#### Access through your organization

Check access to the full text by signing in through your organization.

Access through your organization

**Recommended articles** 

References (0)

Cited by (0)

\* RG and SP contributed equally in preparing the chapter.

View full text

Copyright © 2023 Elsevier Inc. All rights reserved.



All content on this site: Copyright © 2024 Elsevier B.V., its licensors, and contributors. All rights are reserved, including those for text and data mining, AI training, and similar technologies. For all open access content, the Creative Commons licensing terms apply.





Search here...

Login

Register Cart O



Extremophiles: Diversity, Adaptation and Applications



# Applications of Extremophiles

Author(s): Urmimala Sen and Subhra Kanti Mukhopadhyay \* .

**Pp:** 395-417 (23)

DOI: 10.2174/9789815080353122010018

Buy Chapters **30** 

\* (Excluding Mailing and Handling)

#### Abstract

Extremophiles are organisms that can survive in harsh environmental conditions such as varying ranges of temperature, pH, high levels of salinity, extreme pressure and high doses of radiation. They are distributed throughout the Earth's surface and water bodies. They are classified on the basis of their habitats and extreme conditions they inhabit, like oligotrophs, thermophiles, psychrophiles, halophiles, acidophiles, alkaliphiles, piezophiles and radiophiles. Extremophiles have a huge impact on human life. Enzymes obtained from them are nowadays used in industrial microbiology, agriculture, pharmaceuticals and medical diagnostics, bioremediation, and in many more fields. With enormous commercial benefits and advanced scientific techniques, researchers are investigating extremophiles for a better understanding of their metabolism, and survival strategies for newer applications. This chapter focuses on applications of different types of extremophiles in industry, scientific research, medical science, and other fields.

**Keywords:** <u>Acidophiles</u>, <u>Alkaliphiles</u>, <u>Bacteriorhodopsin</u>, <u>Cold-active enzymes</u>, <u>Compatible solutes</u>, <u>Extremophiles</u>, <u>Extremozymes</u>, <u>Halophiles</u>, <u>Heavy-</u> <u>metal resistance</u>, <u>Industrial applications</u>, <u>Mycosporine-like amino acids</u>, <u>Oligotrophs</u>, <u>PGPR</u>, <u>Piezophiles</u>, <u>Polyextermophile</u>, <u>Poly-β hydroxyl</u> <u>alkanoates</u>, <u>Psychrophiles</u>, <u>Radiophiles</u>, <u>Thermophiles</u>.

**Related Journals** 

# Serial No. 425 SPRINGER NATURE Link

**∑** Menu

**Q** Search

Home > Microbial Technology for Health and Environment > Chapter

# Microbial Metabolism of Organophosphates: Key for Developing Smart Bioremediation Process of Next Generation

| Chapter | First Online: 22 March 2020

| pp 361–410 | <u>Cite this chapter</u>



Microbial Technology for Health and Environment

Santanu Pailan, Kriti Sengupta & Pradipta Saha 🖂

Part of the book series: <u>Microorganisms for Sustainability</u> ((MICRO, volume 22))

**679** Accesses **2** <u>Citations</u>

## Abstract

Currently organophosphate compounds constitute one of the largest families of chemical compounds that are used for pest control, mainly for better crop yield worldwide. Due to their toxicity, persistence, and adverse effects, some organophosphates (like parathion and methyl parathion) were classified and registered as extremely hazardous by the World Health Organization (WHO) and US EPA (US Environmental Protection agency) and have been banned in many countries. Some of the hydrolysis intermediates (such as 4–

🔆 Cart

Authors are grateful to SERB, New Delhi, for providing fund to carry out work on organophosphate degradation and to the University of Burdwan, Burdwan, West Bengal. **Author information** 

### **Authors and Affiliations**

Department of Microbiology, Burdwan University, Golapbag, Burdwan, West Bengal, India Santanu Pailan, Kriti Sengupta & Pradipta Saha

## **Corresponding author**

Correspondence to Pradipta Saha.

# **Editor information**

## **Editors and Affiliations**

Department of Microbiology, Babasaheb Bhimrao Ambedkar University, Lucknow, Uttar Pradesh, India Pankaj Kumar Arora

# **Rights and permissions**

Reprints and permissions

# **Copyright information**

© 2020 Springer Nature Singapore Pte Ltd.

# About this chapter

## Cite this chapter

Pailan, S., Sengupta, K., Saha, P. (2020). Microbial Metabolism of Organophosphates: Key for Developing Smart Bioremediation Process of Next Generation. In: Arora, P. (eds)

Microbial Technology for Health and Environment. Microorganisms for Sustainability, vol 22. Springer, Singapore. https://doi.org/10.1007/978-981-15-2679-4\_14

.RIS坐 .ENW坐 .BIB坐

| DOI                       | Published         | Publisher Name             |
|---------------------------|-------------------|----------------------------|
| https://doi.org/10.1007/9 | 22 March 2020     | Springer, Singapore        |
| 78-981-15-2679-4_14       |                   |                            |
|                           |                   |                            |
|                           |                   |                            |
| Print ISBN                | Online ISBN       | eBook Packages             |
| 978-981-15-2678-7         | 978-981-15-2679-4 | <b>Biomedical and Life</b> |

Biomedical and Life Sciences Biomedical and Life Sciences (R0)

## **Publish with us**

Policies and ethics [2]



# Microbial Diversity in Hotspots

2022, Pages 177-205

# Chapter 10 - Bacterial diversity from Garampani warm spring, Assam

Joyasree Das <sup>1</sup>, Pradipta Saha <sup>2</sup>, Srinivasan Krishnamurthi <sup>1</sup>

Show more  $\checkmark$ 

i≡ Outline 🛛 😪 Share 🗦 Cite

https://doi.org/10.1016/B978-0-323-90148-2.00012-2 ス Get rights and content ス

#### Abstract

Prokaryotic organisms constitute one of the major components of biological diversity but in spite of their immense importance, they have not been given enough attention because of their invisibility. Moreover, many lines of evidence suggest that most of them (more than 99%) are not amenable to cultivation. In the present study, bacterial diversity of water of a warm spring located in Assam, a state in the North East of India, was analyzed using a combination of media conditions including nutritionally poor, that is, oligotrophic media. The bacterial representatives were mainly placed in four phyla, namely "Bacteroidetes," "Deinococcus-Thermus," Firmicutes, and Proteobacteria (and classes). Using this approach, three <u>new genera</u> (Aquimonas, Emticicia, Fontibacillus) and three novel species were discovered and few more promising candidates for novel taxa await description. Media biasness was observed with a few novel taxa and species of Emticicia, Flavobacterium, Idonella, Niveibacterium, Rheinheimera, Deinococcus, Pseudomonas, Chitinimonas, Curvibacter, recovered only from the nutritionally poor medium, while taxa belonging to Falsibacillus, Aquimonas, Azoarcus, Comamonas, Pseuaeromonas, Paenibacillus, Fontibacillus, Rubrivivax, and Thauera grew only on nutritionally rich media. Interestingly limited sequencing based on culture-independent 16S rRNA gene-based cloning approach revealed the presence of a large number of

Cyanobacterial sequences and few Proteobacteria clones with a majority of them showing low similarity with cultivated representatives.

### Access through your organization

Check access to the full text by signing in through your organization.

Access through your organization

**Recommended articles** 

References (0)

#### Cited by (0)

View full text

Copyright © 2022 Elsevier Inc. All rights reserved.



All content on this site: Copyright © 2024 Elsevier B.V., its licensors, and contributors. All rights are reserved, including those for text and data mining, AI training, and similar technologies. For all open access content, the Creative Commons licensing terms apply.

■ isbnhelp[at]inflibnet[dot]ac[dot]in

♦ Skip to main content
♦ Skip to Navigation
♦ Screen Reliar

**ISBN** Allotted 1867464

Text Size: A- A A+



Search ISBN

|              | 🗹 Advance Search |              |
|--------------|------------------|--------------|
| Book Title   | Email            | Name of Autł |
| Name of Pub  | Year             | 978-93-8806  |
| Select Produ | Select Langu     |              |

| Search |  |
|--------|--|
|        |  |
|        |  |

Export to Excel

|       |   |   | Search:                             |                |
|-------|---|---|-------------------------------------|----------------|
| #     | Book Title  | ISBN  | Product<br>Form                     | Language       |
| 1     | Hem-Renu  | 978-93-8806<br>72-4                                     | <u> 59-</u>                         | Bengali        |
| nowin | g 1 to 1 of 1 entries   |   |                                     | Previous 1 Nex |
|       | Total Vi  | sitors 533  | 31247                               |                |
|       | Address   | For G   | seneral Quer                        | Y              |
|       | Raja Rammohun Roy ISBN Agen<br>Department of Higher Education<br>Ministry of Education<br>Room No. 13, Jeevan Deep Buildi<br>4th Floor, Parliament Street,<br>New Delhi -110001, India. | cy <b>⊡</b> isbr<br><sup>n,</sup> <b>€</b> +91∙<br>ing, | n-mhrd[at]gov[dot]<br>-11-2334 1739 | in             |

☑ isbnhelp[at]inflibnet[dot]ac[dot]in ► User Manual

For Technical Query Important Links



- ► FAQs
- Contact Us







Search here...

Login

Cart 0



Extremophiles: Diversity, Adaptation and Applications

Register



# Applications of Extremophiles

Author(s): Urmimala Sen and Subhra Kanti Mukhopadhyay \* .

**Pp:** 395-417 (23)

DOI: 10.2174/9789815080353122010018

Buy Chapters **30** 

\* (Excluding Mailing and Handling)

#### Abstract

Extremophiles are organisms that can survive in harsh environmental conditions such as varying ranges of temperature, pH, high levels of salinity, extreme pressure and high doses of radiation. They are distributed throughout the Earth's surface and water bodies. They are classified on the basis of their habitats and extreme conditions they inhabit, like oligotrophs, thermophiles, psychrophiles, halophiles, acidophiles, alkaliphiles, piezophiles and radiophiles. Extremophiles have a huge impact on human life. Enzymes obtained from them are nowadays used in industrial microbiology, agriculture, pharmaceuticals and medical diagnostics, bioremediation, and in many more fields. With enormous commercial benefits and advanced scientific techniques, researchers are investigating extremophiles for a better understanding of their metabolism, and survival strategies for newer applications. This chapter focuses on applications of different types of extremophiles in industry, scientific research, medical science, and other fields.

**Keywords:** <u>Acidophiles</u>, <u>Alkaliphiles</u>, <u>Bacteriorhodopsin</u>, <u>Cold-active enzymes</u>, <u>Compatible solutes</u>, <u>Extremophiles</u>, <u>Extremozymes</u>, <u>Halophiles</u>, <u>Heavy-</u> <u>metal resistance</u>, <u>Industrial applications</u>, <u>Mycosporine-like amino acids</u>, <u>Oligotrophs</u>, <u>PGPR</u>, <u>Piezophiles</u>, <u>Polyextermophile</u>, <u>Poly-β hydroxyl</u> <u>alkanoates</u>, <u>Psychrophiles</u>, <u>Radiophiles</u>, <u>Thermophiles</u>.

**Related Journals** 

| Serial        | No 4                        | 30                                  |         |                        |                            |                   |      |
|---------------|-----------------------------|-------------------------------------|---------|------------------------|----------------------------|-------------------|------|
| Contai        | FUTPLATCE<br>Explore Plus ↔ | Search for products, brands and mor | Q       | Login                  | Become a Seller            | More              | Cart |
| Electronics ~ | TVs & Appliances            | s 🗸 Men 🗸 Women 🗸 Baby & Kids       | s ~ − F | Home & Furniture 🗸 🛛 S | sports, Books & More 🗸 🛛 F | lights Offer Zone |      |

Phalit Nitisastra Vol - 1 By Santosh Kumar Pal (Paperback, Bengali, SANTOS KUMAR PAL)

Price: Not Available

## **Currently Unavailable**

Highlights

Author

Author: SANTOSH KUMAR PAL

SANTOSH KUMAR PAL

358 Pages

Language: Bengali

Publisher: LEVANT BOOKS



For every ₹100 Spent, you earn ○ 2 SuperCoins Max 50 coins per order

#### **Specifications**

| Book            | Phalit Nitisastra Vol - 1 By Santosh Kumar Pal |
|-----------------|--|
| Author          | SANTOSH KUMAR PAL                              |
| Binding         | Paperback                                      |
| Publishing Date | 2021   |
| Publisher       | LEVANT BOOKS                                   |
| Edition         | 3RD EDITION                                    |
| Number of Pages | 358  |
| Language        | Bengali  |



♦ Skip to main content
♦ Skip to Navigation
♦ Screen Reference

**ISBN** Allotted 1867464

Text Size: A- A A+



Search ISBN

|              | 🗹 Advance Search |              |
|--------------|------------------|--------------|
| Book Title   | Email            | Name of Autł |
| Name of Pub  | Year             | 93-88069-73  |
| Select Produ | Select Langu     |              |

| Serial | No. 431 |        |  |
|--------|---------|--------|--|
|        |         | Search |  |
|        |         |        |  |
|        |         |        |  |

Export to Excel

|        |  |                              | Search:                               |                 |
|--------|--|------------------------------|---------------------------------------|-----------------|
| #      | Book Title   | ISBN                         | ₽roduct<br>Form                       | Language        |
| 1      | Falito Nitisastra  | 978-93-880<br>73-1           | 69-                                   | Bengali         |
| Showir | ng 1 to 1 of 1 entries   | Visitors 53                  | 31247                                 | Previous 1 Next |
|        | Address  | For                          | General Query                         | /               |
|        | Raja Rammohun Roy ISBN Age<br>Department of Higher Educat<br>Ministry of Education<br>Room No. 13, Jeevan Deep Bui<br>4th Floor, Parliament Street,<br>New Delhi -110001, India. | ency 🖸 isb<br>ion,<br>Lding, | n-mhrd[at]gov[dot]i<br>1-11-2334 1739 | n               |

For Technical Query Important Links



☑ isbnhelp[at]inflibnet[dot]ac[dot]in ► User Manual



- ► FAQs
- Contact Us

Copyright © 2023 Website Managed and Maintained by INFLIBNET CENTRE, INFOCITY, GANDHINAGAR-382 007, GUJARAT, INDIA.



#### Product details

ASIN : BOCP63C4YY Publisher : LEVANT BOOKS PUBLISHES (1 January 2021) Paperback : 212 pages Reading age : 13 years and up Best Sellers Rank: #1,228,955 in Books (See Top 100 in Books)

Sponsored

#### How would you rate your experience shopping for books on Amazon today?



#### Customer reviews

5 star

No customer reviews



#### You might be interested in



#### **Recently Viewed**



VIEW AL

➡ isbnhelp[at]inflibnet[dot]ac[dot]in

♦ Skip to main content
♦ Skip to Navigation
♦ Screen Reference

Text Size: A- A A+



Search ISBN

| Book Title   | Email        | Name of Autł |
|--------------|--------------|--------------|
|              |              |              |
| Name of Pub  | Year         | 978-81-95198 |
|              |              |              |
| Select Produ | Select Langu |              |

ISBN Allotted 1867464

Export to Excel Search: Product # **Book Title** ISBN Language Form 1 **Dimensions of Bioethics** 978-81-951987-9-Book English 5 Previous 1 Next Showing 1 to 1 of 1 entries 4 **Total Visitors** 5331247 **Address For General Query** isbn-mhrd[at]gov[dot]in Raja Rammohun Roy ISBN Agency Department of Higher Education, **L** +91-11-2334 1739 Ministry of Education Room No. 13, Jeevan Deep Building, 4th Floor, Parliament Street, New Delhi -110001, India. For Technical Query **Important Links** 

isbnhelp[at]inflibnet[dot]ac[dot]in ► User Manual



- ► FAQs
- Contact Us

Copyright © 2023 Website Managed and Maintained by INFLIBNET CENTRE, INFOCITY, GANDHINAGAR-382 007, GUJARAT, INDIA.

■ isbnhelp[at]inflibnet[dot]ac[dot]in

♦ Skip to main content
♦ Skip to Navigation
♦ Screen Reliar

**ISBN** Allotted 1867464

Text Size: A- A A+



Search ISBN

|              | 🗹 Advance Search |              |
|--------------|------------------|--------------|
| Book Title   | Email            | Name of Autł |
| Name of Pub  | Year             | 978-81-95198 |
| Select Produ | Select Langu     |              |

| Serial | No. 434 |        |  |
|--------|---------|--------|--|
|        |         | Search |  |
|        |         |        |  |
|        |         |        |  |

Export to Excel

|        |  | ç                     | Search:         |                 |
|--------|--|-----------------------|-----------------|-----------------|
| #      | Book Title   | ISBN                  | Product<br>Form | ÷<br>Language   |
| 1      | Dimensions of Bioethics  | 978-81-951987-9-<br>5 | Book            | English         |
| Showir | ng 1 to 1 of 1 entries   |                       |                 | Previous 1 Next |
| •      |  |                       |                 |                 |
|        | Total Vis  | sitors 5331247        |                 |                 |
|        | Address  | For Gene              | ral Query       | ,               |
|        | Raja Rammohun Roy ISBN Agenc   | ey 🖂 isbn-mhrc        | l[at]gov[dot]ir | ı               |
|        | Department of Higher Education,<br>Ministry of Education<br>Room No. 13, Jeevan Deep Buildir<br>4th Floor, Parliament Street,<br>New Delbi -110001 India | , +91−11−233          | 4 1739          |                 |

For Technical Query Important Links



☑ isbnhelp[at]inflibnet[dot]ac[dot]in ► User Manual



- ► FAQs
- Contact Us



♦ Skip to main content
♦ Skip to Navigation
♦ Screen Re arr

**ISBN** Allotted 1867464

Text Size: A- A A+



Search ISBN

| To Date      |                  |              |
|--------------|------------------|--------------|
|              | 🗹 Advance Search |              |
| Book Title   | Email            | Name of Autł |
| Name of Pub  | Year             | 978-93-8780  |
| Select Produ | Select Langu     |              |

| Serial | No. 439 |        |  |
|--------|---------|--------|--|
|        |         | Search |  |
|        |         |        |  |
|        |         |        |  |

Export to Excel

|        |   | Search:  |                     |  |
|--------|---|--|---------------------|--|
| #      | Book Title  | Pro<br>ISBN For  | oduct<br>m Language |  |
| 1      | ANIRVAN   | 978-93-87800-<br>39-7                                    | English             |  |
| Showir | ng 1 to 1 of 1 entries  | <b>Visitors</b> 5331247                                  | Previous 1 Next     |  |
|        | Address   | For General (  | Query               |  |
|        | Raja Rammohun Roy ISBN Age<br>Department of Higher Educatio<br>Ministry of Education<br>Room No. 13, Jeevan Deep Buik<br>4th Floor, Parliament Street,<br>New Delhi -110001, India. | ncy 🖂 isbn-mhrd[at]ga<br>on, 🌜 +91-11-2334 1739<br>ding, | ov[dot]in<br>)      |  |

For Technical Query Important Links

☑ isbnhelp[at]inflibnet[dot]ac[dot]in ► User Manual



- ► FAQs
- Contact Us





#### Materials for Biomedical Engineering

Inorganic Micro and Nanostructures

2019, Pages 81-117

Chapter 3 - Structural interpretation, microstructure characterization, mechanical properties, and cytocompatibility study of pure and doped carbonated nanocrystalline hydroxyapatites synthesized by mechanical alloying

Sushovan Lala, Swapan Kumar Pradhan

Materials Science Division, Department of Physics, The University of Burdwan, Burdwan, India

Available online 29 March 2019, Version of Record 29 March 2019.

? What do these dates mean?

Show less 🔨

https://doi.org/10.1016/B978-0-08-102814-8.00004-4 ス Get rights and content ス

#### Abstract

<u>Nanocrystalline</u> biocompatible undoped and Mn-, Mg-, and Zn-doped carbonated hydroxyapatite (HAp) powders have been synthesized via mechanical alloying. A-type <u>carbonation</u> in HAp is confirmed by FTIR analysis. <u>Microstructure characterization</u> in terms of lattice imperfections and phase quantification of the samples are made by analyzing <u>XRD</u> patterns employing Rietveld's method. <u>Microstructure characterization</u> by

#### Chapter

No Image Available

Chapter

### Photonics: A Dream of Modern Technology

By Sourangshu Mukhopadhyay, Shuvra Dey, Subhendu Saha

BookPhotonics and Fiber OpticsEdition1st EditionFirst Published2019ImprintCRC PressPages28eBook ISBN9780429026584



#### ABSTRACT

Optics is established as a potential replacement and alternative of electronic signal in global communication scenarios and in data processing technology because it carries many inherent advantages over its electronic counterpart. These include parallelism, speed, reliability, etc., which are the in-built characters of photons. It is proved that billions of data can be carried by photons at a time. Simultaneously, they can go to the desired destination of millions kilometers apart at a shortest possible time without any difficulty and with a high degree of reliability. They also keep their identity to preserve

the high value of signal-to-noise ratio (SNR). Similarly, the bit error problem is also reduced significantly. The future of communication is believed to be as all photonic communication to exploit multi-level advantages in data transport.

In this chapter, the authors discuss some modern application areas of photonics. The areas are quantum computation, optical soliton, squeezed state of light, photon in quantum cryptography, photon in nano-science and in high-end computation.

| Policies        |  |  |  |
|-----------------|--|--|--|
| Journals        |  |  |  |
| Corporate       |  |  |  |
| Help & Contact  |  |  |  |
| Connect with us |  |  |  |

Registered in England & Wales No. 3099067 5 Howick Place | London | SW1P 1WG

© 2023 Informa UK Limited

RESEARCH ARTICLE | NOVEMBER 05 2020

# Frequency dependent dielectric properties of silver doped hydroxyapatite nanoparticles ≒

Tuli Chatterjee; Amit Kumar Das; Swapan Kumar Pradhan; Ajit Kumar Meikap

Author & Article Information

a) Corresponding author: meikapnitd@yahoo.com
 AIP Conf. Proc. 2265, 030434 (2020)
 https://doi.org/10.1063/5.0017392

Silver doped hydroxyapatite (Ag-HAp (2 mol.%)) nanoparticles of crystallite size 43 nm have been synthesized by employing facile hydrothermal method. The structural, morphological, compositional and electrical characteristics have been studied. The detail dielectric properties at and above room temperature shows a high permittivity value of the synthesized material with a very low dielectric loss which is very much promising for future electronic applications. The high temperature real dielectric response curves are least square fitted with modified Cole-Cole equation to observe the agreement between theoretical and experimental values. The correlated barrier hopping (CBH) model prevails the charge transport mechanism within Ag-HAp on application of external electric field. The plot of frequency dependent imaginary electric modulus of our material shows deviation from ideal Debye-type relaxation nature.

#### Topics

<u>Electrical properties and parameters, Dielectric materials,</u> <u>Dielectric properties, Nanoparticle, Minerals, Transition</u> <u>metals, Charge transport, Chemical synthesis, Doping,</u> <u>Public policy and governance</u>

#### REFERENCES

1. R. U. Mene, M. P. Mahabole, K. C. Mohite and R.

S. Khairnar, Mater. Res. Bull. 50, 227–234 (2014).

https://doi.org/10.1016/j.materresbull.2013.10.040 Google Scholar Crossref

 T. Chatterjee, A. K. Das, S. Lala, S. K. Pradhan and A. K. Meikap, *J. Appl. Phys.* 125, 225107 (1-16) (2019). https://doi.org/10.1063/1.5096452
 Google Scholar Crossref

3. H. H. Adler, *Am. Mineral.* 49, 1002–1015 (1964). Google Scholar

4. D. Ming, J. M. Reau, J. Ravez, J. Gitae and P.
J. Hagenmuller, *J. Solid State Chem.* 116, 185–192 (1995). https://doi.org/10.1006/jssc.1995.1200
Google Scholar Crossref

5. A. K. Jonscher, *Nature* 267, 673–679 (1977). https://doi.org/10.1038/267673a0 Google Scholar Crossref

6. R. Bergman, *J. Appl. Phys.* 88, 1356–1365 (2000). https://doi.org/10.1063/1.373824
Google Scholar Crossref

This content is only available via PDF.

© 2020 Author(s).

You do not currently have access to this content.

#### Sign in

Don't already have an account? Register

#### Sign In

Username

Sign in via your Institution

Password



Reset password

Pay-Per-View Access \$40.00

# Serial No. 450 SPRINGER NATURE Link

**∑** Menu

**Q** Search

ঢ় Cart

Home > ICOL-2019 > Conference paper

# Implementation of Quantum Optical Phase Shift Gate Adopting Multi-passing Technique in Lithium Niobate Based Electro-Optic Crystal

| Conference paper | First Online: 02 March 2021

| pp 687–690 | Cite this conference paper



Minakshi Mandal 🔽 & Sourangshu Mukhopadhyay

Part of the book series: <u>Springer Proceedings in Physics</u> ((volume 258))

**59** Accesses

# Abstract

Quantum phase shift gate operates ordinarily on a single qubit. Here the authors propose a new scheme of implementation of quantum optical phase shift gate adopting multi– passing technique in lithium niobate (LiNbO<sub>3</sub>) based electro-optic crystal. By using the multi-passing technique the phase operation can be done by very low voltage.

The authors acknowledge University Grants Commission (UGC) for the financial support for extending a research fellowship to Minakshi Mandal.

# **Author information**

## **Authors and Affiliations**

Department of Physics, The University of Burdwan, Golapbag, Burdwan, West Bengal, 713104, India Minakshi Mandal & Sourangshu Mukhopadhyay

## **Corresponding author**

Correspondence to Minakshi Mandal.

# **Editor information**

## **Editors and Affiliations**

**IITDTCGHS, B-04, Gyandeep Apartments, Delhi, India** Kehar Singh

Instruments R & D Establishment, Dehradun, India A K Gupta

Instruments R&D Establishment, Dehradun, India Sudhir Khare

Instruments R&D Establishment, Dehradun, India Nimish Dixit

**Optical Engineering, Instrument R&D Establishment, Dehradun, India** Kamal Pant

# **Rights and permissions**

Reprints and permissions

# **Copyright information**

© 2021 The Author(s), under exclusive license to Springer Nature Singapore Pte Ltd.

# About this paper

## Cite this paper

Mandal, M., Mukhopadhyay, S. (2021). Implementation of Quantum Optical Phase Shift Gate Adopting Multi-passing Technique in Lithium Niobate Based Electro-Optic Crystal. In: Singh, K., Gupta, A.K., Khare, S., Dixit, N., Pant, K. (eds) ICOL-2019. Springer Proceedings in Physics, vol 258. Springer, Singapore. https://doi.org/10.1007/978-981-15-9259-1\_158

.RIS ± .ENW ± .BIB ±

| DOI                       | Published         | Publisher Name               |
|---------------------------|-------------------|------------------------------|
| https://doi.org/10.1007/9 | 02 March 2021     | Springer, Singapore          |
| 78-981-15-9259-1_158      |                   |                              |
|                           |                   |                              |
| Print ISBN                | Online ISBN       | eBook Packages               |
| 978-981-15-9258-4         | 978-981-15-9259-1 | <b>Physics and Astronomy</b> |
|                           |                   | <b>Physics and Astronomy</b> |
|                           |                   | <u>(R0)</u>                  |

## **Publish with us**

Policies and ethics [2

# Serial No. 451 SPRINGER NATURE Link

**Ξ** Menu

**Q** Search

🛈 Cart

Home > ICOL-2019 > Conference paper

# Intensity and Voltage Controlled Phase Switching of Light by Joint Effort of Kerr and **Pockels Material**

Conference paper | First Online: 02 March 2021

| pp 691–694 | Cite this conference paper



Suranjan Lakshan Makhopadhyay

Part of the book series: Springer Proceedings in Physics ((volume 258))

ລ 55 Accesses

# Abstract

Propagation of light through a linear-electro optic Pockels material in presence of an external electric field generates a controlled phase shift within it. Similarly Kerr type of electro-optic material causes intensity dependent phase variation. These interesting phenomena of electro-optic material one used in optical communication. Our proposed idea is how one can easily control the phase of an outgoing light wave by controlling suitable biasing voltage as well as the intensity of the input light wave jointly.
**5.** A. Yariv, P. Yeh, *Photonics: Optical Electronics in Modern Communications* (Oxford University Press, 2007)

**Google Scholar** 

 6. S. Lakshan, S. Mukhopadhyay, Intensity and voltage controlled phase switching of light by joint effort of Kerr and Pockels material. Int. Conf. Optics & Electro-optics (19 Oct 2019–22 Oct 2019) IRDE, Dehradun, U.P., India, (2019)

**Google Scholar** 

### Acknowledgements

The authors acknowledge the State Govt. of West Bengal for granting the research fellowship to Suranjan Lakshan.

### **Author information**

#### **Authors and Affiliations**

**Department of Physics, The University of Burdwan, Golapbag, 713104, India** Suranjan Lakshan & Sourangshu Mukhopadhyay

#### **Corresponding author**

Correspondence to Suranjan Lakshan.

### **Editor information**

#### **Editors and Affiliations**

IITDTCGHS, B-04, Gyandeep Apartments, Delhi, India Kehar Singh

Instruments R & D Establishment, Dehradun, India A K Gupta

Instruments R&D Establishment, Dehradun, India



Instruments R&D Establishment, Dehradun, India Nimish Dixit

Optical Engineering, Instrument R&D Establishment, Dehradun, India Kamal Pant **Rights and permissions** 

**Reprints and permissions** 

### **Copyright information**

© 2021 The Author(s), under exclusive license to Springer Nature Singapore Pte Ltd.

### About this paper

#### **Cite this paper**

Lakshan, S., Mukhopadhyay, S. (2021). Intensity and Voltage Controlled Phase Switching of Light by Joint Effort of Kerr and Pockels Material. In: Singh, K., Gupta, A.K., Khare, S., Dixit, N., Pant, K. (eds) ICOL-2019. Springer Proceedings in Physics, vol 258. Springer, Singapore. https://doi.org/10.1007/978-981-15-9259-1\_159

.RIS ± .ENW ± .BIB ±

DOIPublishedPublisher Namehttps://doi.org/10.1007/902 March 2021Springer, Singapore78-981-15-9259-1\_159Online ISBNeBook PackagesPrint ISBN0nline ISBNeBook Packages978-981-15-9258-4978-981-15-9259-1Physics and Astronomy<br/>Physics and Astronomy<br/>(R0)



6085dc0dea3e&context=ubx)

< Photonics, Plasmonics and Information Optics (https://www.taylorfrancis.com/books/mono/10.1201/9781003047193/photonics-plasmonics-information-optics?refid=2f10464b-1436-4c98-bd73-</p>

Show Path 🗸

Chapter



# Investigation on Some Fast Optical/Opto-Electronic Switching Systems for Implementing Different Modulation Schemes

By Minakshi Mandal (/search?contributorName=Minakshi Mandal&contributorRole=author&redirectFromPDP=true&context=ubx), Suranjan Lakshan (/search? contributorName=Suranjan Lakshan&contributorRole=author&redirectFromPDP=true&context=ubx), Debashri Saha (/search?contributorName=Debashri Saha&contributorRole=author&redirectFromPDP=true&context=ubx), Agnijita Chatterjee (/search?contributorName=Agnijita Chatterjee&contributorRole=author&redirectFromPDP=true&context=ubx), Sourangshu Mukhopadhyay (/search?contributorName=Sourangshu Mukhopadhyay&contributorRole=author&redirectFromPDP=true&context=ubx)

Book Photonics, Plasmonics and Information Optics (https://www.taylorfrancis.com/books/mono/10.1201/9781003047193/photonics-plasmonicsinformation-optics?refld=db2db8df-312e-409a-9e4a-7250ba1fcebe&context=ubx)

| Edition         | 1st Edition   |
|-----------------|---------------|
| First Published | 2021          |
| Imprint         | CRC Press     |
| Pages           | 33            |
| eBook ISBN      | 9781003047193 |
|                 |               |

< Share

#### ABSTRACT

< Previous Chapter (chapters/edit/10.1201/9781003047193-4/design-superlens-using-2d-photonic-crystal-various-geometries-polarized-incidence-moumitabanerjee-pampa-debnath-arpan-deyasi?context=ubx)

Next Chapter > (chapters/edit/10.1201/9781003047193-6/slotted-photonic-crystal-waveguide-mrinal-sen-tanmoy-datta?context=ubx)



♦ Skip to main content
♦ Skip to Navigation
♦ Screen Reference

**ISBN** Allotted 1867464

Text Size: A- A A+



Search ISBN

|              | 🗹 Advance Search |              |
|--------------|------------------|--------------|
| Book Title   | Email            | Name of Autł |
| Name of Pub  | Year             | 978-81-9553  |
| Select Produ | Select Langu     |              |

| Serial | No. 453 |        |  |
|--------|---------|--------|--|
|        |         | Search |  |
|        |         |        |  |
|        |         |        |  |

Search: Product # **Book Title** ISBN Form Language 1 Proceedings of the National 978-81-955389-Digital English download and Conference on Materials, Devices 3-5 and Circuits in Communication online Technology, 2020 (MDCCT 2020) (Edition -1) Previous 1 Next Showing 1 to 1 of 1 entries 

Export to Excel



Raja Rammohun Roy ISBN Agency Department of Higher Education, Ministry of Education Room No. 13, Jeevan Deep Building, 4th Floor, Parliament Street, New Delhi -110001, India.

- ☑ isbn-mhrd[at]gov[dot]in
- **L** +91-11-2334 1739

For Technical Query

**Important Links** 

user Manual User Manual

**L** +91-79-2326 8229

- FAQs
- Contact Us

Copyright © 2023 Website Managed and Maintained by INFLIBNET CENTRE, INFOCITY, GANDHINAGAR-382 007, GUJARAT, INDIA.



#### A new scheme of conducting amplitude modulation with KDP crystal using multiple-number of message signals on a single carrier light beam followed by different analyzer

Minakshi Mandal and Sourangshu Mukhopadhyay

Department of Physics, The University of Burdwan, Golapbag, Burdwan, 713104, West Bengal, India Email: minakshi.bcc@gmail.com

#### Abstract

Electro-optic Modulators have several applications like data transmission, data processing, all-optical switches in modern communication system. Non-linear materials like lithium niobate (LiNbO<sub>3</sub>), potassium dihydrogen phosphate (KDP), lithium tantalate (LiTaO<sub>3</sub>) are basically used to design the Pockels type of electro-optic modulator. Using the non-linear property of the electro-optic material with the application of suitable biasing voltage, amplitude modulation, phase modulation, polarization modulation, frequency modulation of carrier light signal are conducted. Here the authors proposed a new scheme where the amplitude modulation with KDP crystal is conducted by using multiple number of message signals on a single carrier light beam.

#### **INTRODUCTION**

Electro-optic materials are established as high speed electro-optic devices due to their non-linear optical property. Amplitude modulation, phase modulation, polarization modulation, frequency modulation etc. are occurred due this non-linear property of the electro-optic material which leads to develop the high speed communication devices [1]. Sharp variation of transmission factor of KDP crystal by super parabolic biasing voltage is derived using the amplitude modulation scheme[2]. An investigation is done to achieve the highest phase difference between the orthogonal components of light in lithium niobate based electro-optic system [3]. Increase of sideband powers in parallel phase modulation is done in lithium niobate based electro-optic system [4]. Highest Transmission Factor is obtained in Case of KDP Based Electro-Optic Crystal by the Adjustment of Suitable Biasing Voltage and Number of Feedback Passing [5]. A systematic and quantitative analysis of residual amplitude modulation is derived from the birefringence character of the electro-optic crystal [6]. In this paper, the author conducted a new scheme of amplitude modulation with KDP crystal where multiple number of message signals are used in a single carrier light beam.

Here the novelty of the proposed method is that putting the analyzer in different suitable positions, one can obtain output modulated beam containing different message signals.

#### AMPLITUDE MODULATION BY ELECTRO-OPTIC MODULATOR

Amplitude modulation is used to transmit information through radio carrier wave in electronic communication system. Phase and polarization states are the fundamental modulation taken place in the electro-optic modulator. These modulations lead to the amplitude modulation in desired conditions. When a light polarized in x' direction passes through the KDP crystal of length (*l*) along Z direction with the application of an external electric field (E) parallel to Z direction (Fig. 1), then the refractive index of the material is expressed as  $n_{x'} = n_0 - \frac{1}{2}n_0^3 r_{63} El$ . Now the output light signal is expressed as  $E_{x'}(l) =$ 

$$E_{x'}(0) \exp i \left( \omega t - kn_0 l + \frac{1}{2} kn_0^3 r_{63} El \right).$$

When the polarization direction of the light beam is changed to y' direction, then the refractive index of the material is expressed as,  $n_{y'} = n_0 + \frac{1}{2}n_0^3 r_{63} El$ . Now the output light signal is expressed as  $E_{y'}(l) = E_{y'}(0) \exp i$  ( $\omega t - kn_0l + \frac{1}{2}kn_0^3 r_{63} El$ ). Now these two light beams are taken from a single light source, polarized in 45 to both x' and y' and passes through the KDP crystal, then output signal passes through an analyzer which is perpendicular to input polarization state, then the output signal emerging from the analyzer is an amplitude modulated output signal. The expression of the amplitude modulated signal is



Figure 1. Amplitude Modulation in KDP Crystal

 $E'(l) = \frac{4}{2} \left[ \exp i \left( \omega t - kn_0 l + \frac{1}{2} kn_0^3 r_{63} V \right) - \exp i \left( \omega t - kn_0 l - \frac{1}{2} kn_0^3 r_{63} V \right); \quad \text{[where } V = El \text{]}$ The intensity of the amplitude modulated signal is expressed as

Serial No. 454 isbnhelp[at]inflibnet[dot]ac[dot]in

♦ Skip to main content
♦ Skip to Navigation
♦ Screen Reliar

**ISBN** Allotted 1867464

Text Size: A- A A+



Search ISBN

|              | 🗹 Advance Search |              |
|--------------|------------------|--------------|
| Book Title   | Email            | Name of Autł |
| Name of Pub  | Year             | 978-81-9553  |
| Select Produ | Select Langu     |              |

| Serial | No. 454 |        |  |
|--------|---------|--------|--|
|        |         | Search |  |
|        |         |        |  |
|        |         |        |  |

Export to Excel

Search:

Product # **Book Title** ISBN Form Language 1 Proceedings of the National 978-81-955389-Digital English Conference on Materials, Devices 3-5 download and and Circuits in Communication online Technology, 2020 (MDCCT 2020) (Edition -1) Previous 1 Next Showing 1 to 1 of 1 entries 





For Technical Query

**Important Links** 

**L** +91-79-2326 8229

- FAQs
- Contact Us

Copyright © 2023 Website Managed and Maintained by INFLIBNET CENTRE, INFOCITY, GANDHINAGAR-382 007, GUJARAT, INDIA.



#### An all-optical scheme for developing a programmable and integrated Pauli X, Y & Z gate based logic system using phase and frequency encoding principles

Snigdha Hazra and Sourangshu Mukhopadhyay

Department of Physics, The University of Burdwan, Burdwan-713104, West Bengal, India E-mail: snigdhahazra29@gmail.com

#### Abstract

The heart of a quantum computer is made of quantum logic blocks that consists of a programmed sequence of quantum logic gates. In this paper an all-optical scheme for developing a programmable and integrated Pauli X, Y & Z gate based logic system is proposed. The system is developed by using phase encoding and frequency encoding techniques jointly. Since the proposed scheme is all-optical in nature, the system exhibits high degree of parallelism and high speed of operation.

#### **INTRODUCTION**

Quantum behavior of light plays a potential and significant role in computation and communication. Quantum logic is based on the principles of quantum mechanics. The essential components of a quantum computer are quantum logic gates that operate on qubits. High degree of parallelism, enormous storage capacity, superfast speed of operation, low noise, no cross-talk, data security etc. can be achieved by quantum computation. Various quantum logic gates such as Pauli X, Y, Z gates, SRZ gate, SRCZ gate, NOT gate, CNOT gate, Fredkin gate, Hadamard gate, Toffoli gate etc. were implemented in last few years by several ways. Optics is found as a very suitable and potential candidate in quantum computation. In quantum computer, the basic unit of information is qubit or quantum bits. Different physical properties of light (Intensity, Polarization, Phase, Frequency etc.) are used to encode the bits. Several all-optical integrated logic systems are developed earlier using different encoding techniques. All-optical integrated Pauli X, Y & Z gates were implemented using frequency encoding technique by Sarkar et.al.<sup>1</sup>.The integrated square root of Pauli Z (SRZ) gate were also developed using polarization and phase encoding techniques jointly<sup>2</sup>. Integrated Pauli X, Y & Z quantum gates with optical switches were also implemented previously<sup>3</sup>.

In this paper, an all-optical scheme for developing a programmable and integrated Pauli X, Y & Z gate based logic system using phase and frequency encoding principles of light is proposed. Pauli X, Y & Z gates are very much advantageous in the world of quantum computing from application point of view. They act on single qubit states. The general qubit states can be represented as,  $|\Psi\rangle = a|0\rangle + b|1\rangle$  Where a and b are amplitude coefficients. The states  $|0\rangle$  and  $|1\rangle$  are represented by  $2\times 1$  column matrices.

$$|0> = \begin{pmatrix} 1\\ 0 \end{pmatrix} \& |1> = \begin{pmatrix} 0\\ 1 \end{pmatrix}$$

The Pauli-X gate is represented by Pauli X matrix.

 $X = \begin{pmatrix} 0 & 1 \\ 1 & 0 \end{pmatrix}$ Hence, when it operates on |0 > and |1 > respectively, we get,

$$\begin{pmatrix} 0 & 1 \\ 1 & 0 \end{pmatrix} \begin{pmatrix} 1 \\ 0 \end{pmatrix} = \begin{pmatrix} 0 \\ 1 \end{pmatrix} \qquad \begin{pmatrix} 0 & 1 \\ 1 & 0 \end{pmatrix} \begin{pmatrix} 0 \\ 1 \end{pmatrix} = \begin{pmatrix} 1 \\ 0 \end{pmatrix}$$

This shows that the Pauli-X gate acts as a NOT gate. The gate is reversible in nature as we get back the original state by repeating the operation.

Pauli-Y gate is represented by Pauli Y matrix.

$$Y = \begin{pmatrix} 0 - i \\ i & 0 \end{pmatrix}$$

Hence, when it operates on  $|0\rangle$  and  $|1\rangle$  respectively, we get,

$$\begin{pmatrix} 0 & -i \\ i & 0 \end{pmatrix} \begin{pmatrix} 1 \\ 0 \end{pmatrix} = \begin{pmatrix} 0 \\ i \end{pmatrix} \qquad \begin{pmatrix} 0 & -i \\ i & 0 \end{pmatrix} \begin{pmatrix} 0 \\ 1 \end{pmatrix} = \begin{pmatrix} -i \\ 0 \end{pmatrix}$$

Again,

$$\begin{pmatrix} 0 - i \\ i & 0 \end{pmatrix} \begin{pmatrix} 0 \\ i \end{pmatrix} = \begin{pmatrix} 1 \\ 0 \end{pmatrix} \qquad \begin{pmatrix} 0 - i \\ i & 0 \end{pmatrix} \begin{pmatrix} -i \\ 0 \end{pmatrix} = \begin{pmatrix} 0 \\ 1 \end{pmatrix}$$

Pauli-Y gate also follows reversible character. By repeating the operation, the original state is obtained. Pauli-Z gate is represented by Pauli-Z matrix.

$$Z = \begin{pmatrix} 1 & 0 \\ 0 - 1 \end{pmatrix}$$

■ isbnhelp[at]inflibnet[dot]ac[dot]in

♦ Skip to main content
♦ Skip to Navigation
♦ Screen Reliar

**ISBN** Allotted 1867464

Text Size: A- A A+



### Search ISBN

|              | 🗹 Advance Search |              |
|--------------|------------------|--------------|
| Book Title   | Email            | Name of Autł |
| Name of Pub  | Year             | 978-81-9553  |
| Select Produ | Select Langu     |              |

Search

Export to Excel

Search:

| #      | <b>Book Title</b>   | ISBN                  | Product<br>Form                   | Language        |
|--------|---|-----------------------|-----------------------------------|-----------------|
| 1      | Proceedings of the National<br>Conference on Materials, Devices<br>and Circuits in Communication<br>Technology, 2020 (MDCCT 2020)<br>(Edition -1) | 978-81-955389-<br>3-5 | Digital<br>download and<br>online | English         |
| Showir | ng 1 to 1 of 1 entries  |                       |                                   | Previous 1 Next |
|        |   |                       |                                   |                 |



AddressFor General QueryRaja Rammohun Roy ISBN Agency<br/>Department of Higher Education,<br/>Ministry of Education<br/>Room No. 13, Jeevan Deep Building,<br/>4th Floor, Parliament Street,<br/>New Delhi -110001, India.☑ isbn-mhrd[at]gov[dot]in<br/>
<br/>
<br/>
<br/>
<br/>
<br/>
<br/>

For Technical Query

**Important Links** 

Lat]inflibnet[dot]ac[dot]in ► User Manual

**\$** +91-79-2326 8229

- FAQs
- Contact Us

Copyright © 2023 Website Managed and Maintained by INFLIBNET CENTRE, INFOCITY, GANDHINAGAR-382 007, GUJARAT, INDIA.



#### An alternative proposal for implementation of quantum optical oscillator with tristate Pauli-Z gate

Mir Nadim Sarfaraj and Sourangshu Mukhopadhyay

<sup>1</sup>Dept. of Physics, The University of Burdwan, Golapbag, Burdwan, West Bengal – 713104, India Email: mirnadim222@gmail.com

#### Abstract

In quantum optical computation, a pair of states are required to be appeared alternatively at the output of a quantum optical oscillator at a certain time interval. A two-state quantum optical oscillator has been developed earlier with oscillating intensity of light. In this paper authors have a new proposal for implementation of quantum optical oscillator using tristate Pauli-Z quantum gate, with oscillating phase of the light signal.

#### INTRODUCTION

An oscillator can deliver output signal even without any external input signal. It can produce periodic wave of a desired frequency at its output. In quantum optical computing, light is used as the signal carrier [1]. So, at the output of a quantum optical oscillator circuit one can expect the oscillation of one of the parameters of light. These parameters can be intensity, phase, etc. Intensity oscillation circuit [2] has been proposed earlier by Shuvra Dey et al. In this paper authors are going to present a new proposal for implementation of quantum oscillator circuit with tristate Pauli-Z logic gate [3]. Here, the oscillating parameter is the phase of the light signal. All operations are done by using intensity and phase modulation of light signals [4]. As the phase of light has quantum behaviour, at the output of the oscillator the quantum nature of the states will be maintained.

#### TRISTATE PAULI-Z GATE AND ITS OPERATION

There are two different circuits corresponding to tristate Pauli-Z gate [3]. One of those is used here to design tristate oscillator circuit. The circuit diagram of tristate Pauli-Z gate is shown in Figure 1.



Figure1. Circuit diagram of Pauli-Z gate.

The circuit of Pauli-Z gate gives the output  $\begin{pmatrix} C_0 \\ -C_1 \\ -C_2 \end{pmatrix}$  for the input  $\begin{pmatrix} C_0 \\ C_1 \\ C_2 \end{pmatrix}$ . The gate matrix corresponding to this circuit is  $Z = \begin{pmatrix} 1 & 0 & 0 \\ 0 & -1 & 0 \\ 0 & 0 & -1 \end{pmatrix}$ .

#### PROPOSED SCHEME FOR IMPLEMENTATION OF PAULI-Z GATE BASED TRISTATE OSCILLATOR WITH LIGHT

The optical circuit diagram for quantum optical oscillator circuit using tristate Pauli-Z gate is shown in Figure 2. The input signals are  $C_0$ ,  $C_1$  and  $C_2$  and the corresponding outputs are taken at  $O_1$ ,  $O_2$  and  $O_3$  respectively. Feedbacks to the inputs are taken from respective outputs in all three channels via EDFAs (Erbium-doped Fibre Amplifier) and electro-optical modulators (EOMs). Signal is applied to the inputs for once and then they are removed to receive the oscillatory operations.

C Photonics and Fiber Optics (https://www.taylorfrancis.com/books/mono/10.1201/9780429026584/photonics-fiber-optics?refid=474d1932-4137-48a6-8772-54d6c4d3990c&context=ubx)

Chapter



# Nonlinear Frequency Conversion—A Versatile Approach to Achieve Broadly Tunable Coherent Radiation

By Udit Chatterjee (/search?contributorName=Udit Chatterjee&contributorRole=author&redirectFromPDP=true&context=ubx)

| Book <u>Photonics and</u> | d Fiber Optics (https://www.taylorfrancis.com/books/mono/10.1201/9780429026584/photonics-fiber-optics?refld=a8a788bb-a885- |
|---------------------------|--|
| <u>4218-ab97-d4</u>       | <u>18a7851b030&amp;context=ubx)</u>  |
| Edition                   | 1st Edition  |
| First Published           | 2019   |
| Imprint                   | CRC Press  |
| Pages                     | 27   |
| eBook ISBN                | 9780429026584  |
|                           |  |
| 😋 Share                   |  |

#### ABSTRACT

<
 Previous Chapter (chapters/edit/10.1201/9780429026584-7/fluorescence-fundamentals-applications-mohan-singh-mehata?context=ubx)
 Next Chapter > (chapters/edit/10.1201/9780429026584-9/photonics-application-high-voltage-engineering-nirmal-kumar-roy?context=ubx)

 $\sim$ 

ELECTRICAL ENGINEERING DEVELOPMENTS

# Phase-Locked Loops

# Structure, Functions and Applications





# **Serial No. 457** Finase-Locked Loops. Surdeure, Functions and Applications

\$230.00

#### Shambhu N Sharma, PhD (Editor)

Professor of Electrical Engineering and Former Head, Electrical Engineering Department, Sardar Vallabhbhai National Institute of Technology, Surat, India

Series: Electrical Engineering Developments BISAC: TEC007000

The historic account of the Phase-Locked Loops can be traced back from the idea of designing an electromechanical system with the objective of controlling the oscillation of the pendulum of the bell Great George. The method is to contrast the phase of pendulum and the incoming telegraph signal phase using the electromechanical system. That generates the correction signal varying the pendulum oscillation. The idea was conceived as well as implemented by David Robertson, Professor of Electrical Engineering at the University of Bristol.

The term Phase-Locked Loop was coined to this technique by later Researchers in 1932. Professor David Robertson is credited to the Phase-Locked Loop for pioneering the technique. In general setting, the Phase-Locked Loops are for synchronization purposes. The phase locked loops perspective hinges on the analysis, functions and applications.

Add to Wishlist

| Binding    |                         |
|------------|-------------------------|
| Hardcove   | er 🗸 🖌 Clear            |
| Publicatio | on Date: September 2020 |
| Status: Av | railable                |
| Pages: 34  | 5                       |
|            |                         |
|            |                         |
| 1          | Add to cart             |
|            |                         |
|            |                         |
|            |                         |

Engineering Tags: 978-1-53618-338-2, 978-1-53618-490-7, electrical engineering

Like 1

|  | Publish with Us | Additional information | Table of Contents |  |
|--|-----------------|------------------------|-------------------|--|
|--|-----------------|------------------------|-------------------|--|

#### Table of Contents

#### Table of Contents

Preface

Chapter 1. A New Phase-Locked Loop structure in Hybrid CMOS-Memristor Technology (Ashok Srivastava, Naheem Olakunle Adesina and Yang Liu, Division of Electrical and Computer Engineering, Louisiana State University, Baton Rouge, LA, US, and others)

Chapter 2. Biological Synchronization Modeled by PLLs and Pitch Detection (Patricia A. Pelle and Claudio F. Estienne, Universidad de Buenos Aires, Facultad de Ingenieria, Instituto de Ingenieria Biomedica, Buenos Aires, Argentina)

Chapter 3. Synchronous Auditory Representation of Speech Modeled with PLLS (Patricia A. Pelle and Claudio F. Estienne, Universidad de Buenos Aires.Facultad de Ingenieria. Instituto de Ingenieria Biomedica, Buenos Aires, Argentina)

Chapter 4. Nonlinear Dynamics of Time-Delay Digital Tanlock Loop (Bishwajit Paul and Tanmoy Banerjee, Department of Physics, Krishnanagar Government College, Nadia, West Bengal, and others)

Chapter 5. Carleman Linearization of a Stochastic Phase-Locked Loop System (Prashant G. Medewar, S. N. Sharma and Hirenkumar G. Patel, Department of Electrical Engineering, SardarVallabhbhai National Institute of Technology, Surat, Gujarat, India)

Chapter 6. An Application of Phase-Locked Loop in Robust Control Design (Bhanu Pratap and Akhilesh Swarup, Department of Electrical Engineering, National Institute J. Technology Kurukshetra, Haryana, India)

(Sonam Kharade, Sushama Wagh and Navdeep Singh, Electrical Engineering Department, Veermata Jijabai Technological Institute, Mumbai, Maharashtra, India)

Chapter 8. A Sensitivity Analysis of Phase- Locked Loop Systems (Manish Yadav, Prashant G. Medewar, Arvind K. Singh and Hirenkumar G. Patel, Electrical Engineering Department, Saradar Vallabhbhai National Institute of Technology, Gujrat, Surat, India, and others)

Chapter 9. Phase-Locked Loops for UPQC in Distribution System along with Optimized PI-Controller Gains (Sabha Raj Arya and Sayed Javed Alam, Department of Electrical Engineering, Sardar Vallabhai National Institute of Technology, Surat, India)

Index

#### Nova Science Publishers, Inc.

Nova publishes a wide array of books and journals from authors around the globe, focusing on Medicine and Health, Science and Technology and the Social Sciences and Humanities.

We publish over 1,500 new titles per year by leading researchers each year, and have a network of expert authors, editors and advisors spanning the global academic community in pursuit of advanced research developments.

Headquarters: Nova Science Publishers, Inc. 415 Oser Avenue, Suite N Hauppauge, NY, 11788 USA

Tel: 1-631-231-7269 Fax: 1-631-231-8175



I would like to receive emails from Nova Science Publishers. Sign me up!

# Serial No. 458 SPRINGER NATURE Link

**∑** Menu

**Q** Search

Home > Cybersecurity > Chapter

## Quantum Oscillations: A Promising Field for Secure Communication

| Chapter | First Online: 26 March 2022

| pp 69-81 | Cite this chapter



Tanmoy Banerjee 🔽 & Biswabibek Bandyopadhyay

Part of the book series: <u>Studies in Big Data</u> ((SBD,volume 102))

**618** Accesses

### Abstract

Quantum communication has been identified as the most secured technique of future communication systems. Its security is directly tied up to the fundamental laws of quantum mechanics that never fail. The success of quantum communication hinges on the notion of oscillations and synchronization in the quantum regime. In this chapter, we discuss self-sustained oscillations and their synchronizations in the quantum regime. Unlike classical oscillations, quantum oscillations are bounded by several constraints of quantum mechanics. For example, although a prominent phase trajectory is possible in

🗘 Cart

#### Serial No. 458 Author information

#### **Authors and Affiliations**

Chaos and Complex Systems Research Laboratory, Department of Physics, University of Burdwan, Burdwan, 713 104, India Tanmoy Banerjee & Biswabibek Bandyopadhyay

#### **Corresponding author**

Correspondence to Tanmoy Banerjee.

### **Editor information**

#### **Editors and Affiliations**

Department of Mathematics and Computer Science, Faculty of Science, Menoufia University, Shibin El Kom, Egypt Ahmed A. Abd El-Latif

**Department of Physics, Aristotle University of Thessaloniki, Thessaloniki, Greece** Christos Volos

### **Rights and permissions**

**Reprints and permissions** 

# **Copyright information**

© 2022 The Author(s), under exclusive license to Springer Nature Switzerland AG

### About this chapter

#### Cite this chapter

Banerjee, T., Bandyopadhyay, B. (2022). Quantum Oscillations: A Promising Field for Secure Communication. In: Abd El-Latif, A.A., Volos, C. (eds) Cybersecurity. Studies in Big

Data, vol 102. Springer, Cham. https://doi.org/10.1007/978-3-030-92166-8\_4

.RIS ± .ENW ± .BIB ±

DOI https://doi.org/10.1007/9 78-3-030-92166-8\_4 Published 26 March 2022 Publisher Name Springer, Cham

Print ISBN 978-3-030-92165-1 Online ISBN 978-3-030-92166-8 eBook Packages Intelligent Technologies and Robotics Intelligent Technologies and Robotics (R0)

#### **Publish with us**

Policies and ethics [2

# Serial No. 459 SPRINGER NATURE Link

 $\equiv$  Menu

**Q** Search

Home > Advanced Oxidation Processes in Dye-Containing Wastewater > Chapter

# Metal Oxide Heterostructured Nanocomposites for Wastewater Treatment

| Chapter | First Online: 12 May 2022

| pp 267-303 | Cite this chapter



Advanced Oxidation Processes in

**Dye-Containing Wastewater** 

M. Mondal, M. Ghosh, H. Dutta & S. K. Pradhan 🖂

Part of the book series: <u>Sustainable Textiles: Production, Processing, Manufacturing &</u> <u>Chemistry</u> ((STPPMC))

**329** Accesses

### Abstract

Over the past few decades, nanocrystalline metal oxide semiconductor-based photocatalytic technology has drawn significant attention for reducing the recent energy crisis by converting solar energy into potential energies and remediating environmental pollution. Two traditional semiconductors, TiO<sub>2</sub> and ZnO, have been widely investigated as a photocatalyst for water purification among several metal oxide semiconductors. However, the photocatalytic performance of the above-mentioned metal oxide-based

. Cart

research facilities in the Dept. of Physics of the University of Burdwan through CAS–I, CAS–II and FIST programs, respectively. M. Mondal wishes to thank the Govt. of West Bengal for financial support through the Vivekananda Merit–Cum–Means fellowship for non–NET fellows.

### **Author information**

#### **Authors and Affiliations**

Materials Science Division, Department of Physics, The University of Burdwan, Golapbag, Burdwan, West Bengal, 713104, India M. Mondal, M. Ghosh & S. K. Pradhan

Department of Physics, Vivekananda Mahavidyalaya, Burdwan, West Bengal, 713104, India H. Dutta

II. Dutta

### **Corresponding author**

Correspondence to <u>S. K. Pradhan</u>.

# **Editor information**

### **Editors and Affiliations**

**SgT Group & API, Hong Kong, Kowloon, Hong Kong** Subramanian Senthilkannan Muthu

#### Western University, London Ontario, Canada Ali Khadir

# **Rights and permissions**

Reprints and permissions

### **Copyright information**

© 2022 The Author(s), under exclusive license to Springer Nature Singapore Pte Ltd.

#### Serial No. 459 About this chapter

#### Cite this chapter

Mondal, M., Ghosh, M., Dutta, H., Pradhan, S.K. (2022). Metal Oxide Heterostructured Nanocomposites for Wastewater Treatment. In: Muthu, S.S., Khadir, A. (eds) Advanced Oxidation Processes in Dye–Containing Wastewater. Sustainable Textiles: Production, Processing, Manufacturing & Chemistry. Springer, Singapore. https://doi.org/10.1007/978-981-19-0987-0\_12

#### <u>.RIS</u> <u>★</u> <u>.ENW</u> <u>★</u> <u>.BIB</u> <u>↓</u>

| DOI                       | Published         | Publisher Name                 |
|---------------------------|-------------------|--------------------------------|
| https://doi.org/10.1007/9 | 12 May 2022       | Springer, Singapore            |
| 78-981-19-0987-0_12       |                   |                                |
| Print ISBN                | Online ISBN       | eBook Packages                 |
| 978-981-19-0986-3         | 978-981-19-0987-0 | <b>Chemistry and Materials</b> |
|                           |                   | Science                        |
|                           |                   | <b>Chemistry and Material</b>  |
|                           |                   | Science (R0)                   |

#### **Publish with us**

Policies and ethics 🖸

↓ Skip to main content
 ↓ Skip to Navigation
 ↓ Screen Re error

**ISBN** Allotted 1867464

Text Size: A- A A+



Search ISBN

|              | 🗹 Advance Search |              |
|--------------|------------------|--------------|
| Book Title   | Email            | Name of Autł |
| Name of Pub  | Year             | 978-81-9553  |
|              |                  |              |
| Select Produ | Select Langu     |              |

Search

Export to Excel

Search:

| #      | <b>Book Title</b>   | ISBN                  | Product<br>Form                   | Language        |
|--------|---|-----------------------|-----------------------------------|-----------------|
| 1      | Proceedings of the National<br>Conference on Materials, Devices<br>and Circuits in Communication<br>Technology, 2020 (MDCCT 2020)<br>(Edition -1) | 978-81-955389-<br>3-5 | Digital<br>download and<br>online | English         |
| Showir | ng 1 to 1 of 1 entries  |                       |                                   | Previous 1 Next |
|        |   |                       |                                   |                 |



AddressFor General QueryRaja Rammohun Roy ISBN Agency<br/>Department of Higher Education,<br/>Ministry of Education<br/>Room No. 13, Jeevan Deep Building,<br/>4th Floor, Parliament Street,<br/>New Delhi -110001, India.Image: Constant of the street of the st

For Technical Query

**Important Links** 

**L** +91-79-2326 8229

- FAQs
- Contact Us

Copyright © 2023 Website Managed and Maintained by INFLIBNET CENTRE, INFOCITY, GANDHINAGAR-382 007, GUJARAT, INDIA.



| Citations (0) | References (0)       |   |
|---------------|----------------------|---|
| Rese          | archGate has not bee | n able to resolve any citations for this publication. |
|               |                      |   |



**Business solutions** 

Advertising

Recruiting



Get IT ON Google Play

| Company        | Support     |
|----------------|-------------|
| About us       | Help Center |
| News           |             |
| <u>Careers</u> |             |
|                |             |

© 2008-2024 ResearchGate GmbH. All rights reserved.

 $Terms \cdot Privacy \cdot Copyright \cdot Imprint \cdot Consent preferences$ 

#### Serial No. 461 elp[at]inflibnet[dot]ac[dot]in

**ISBN** Allotted 1867464

♦ Skip to main content
♦ Skip to Navigation
♦ Screen Reference

Text Size: A- A A+



Search ISBN

|              | 🗹 Advance Search |              |
|--------------|------------------|--------------|
| Book Title   | Email            | Name of Autł |
| Name of Pub  | Year             | 978-81-9553  |
| Select Produ | Select Langu     |              |

| Serial | No. 461 |        |  |
|--------|---------|--------|--|
|        |         | Search |  |
|        |         |        |  |
|        |         |        |  |

Export to Excel

Search: Product # **Book Title** ISBN Form Language 1 Proceedings of the National 978-81-955389-Digital English download and Conference on Materials, Devices 3-5 and Circuits in Communication online Technology, 2020 (MDCCT 2020) (Edition -1) Previous 1 Next Showing 1 to 1 of 1 entries 





For Technical Query

**Important Links** 

**L** +91-79-2326 8229

- ► FAQs
- Contact Us

Copyright © 2023 Website Managed and Maintained by CENTRE, INFOCITY, GANDHINAGAR-382 007, GUJARAT, INDIA.



#### Structural and AC conductivity analysis of Gd<sub>2</sub>NiTiO<sub>6</sub> double Perovskite material

Sujan Malik<sup>1</sup> and Abhigyan Dutta<sup>2</sup>

Department of Physics, The University of Burdwan, Burdwan-713104, India. Email: adutta@phys.buruniv.ac.in<sup>1</sup>, sujanmalikphysics@gmail.com<sup>2</sup>

#### Abstract

This study investigates the structural and electrical properties of double perovskite Gd<sub>2</sub>NiTiO<sub>6</sub> (GNT) prepared via the sol-gel combustion method. Rietveld refinement of the XRD pattern confirms the monoclinic structure with space group P2<sub>1</sub>/n. The crystallite size of the composition sintered at temperature 1200 °C was found to be ~ 47.14 nm. The electrical study established the dc conductivity to be thermally activated and indicated a negative temperature coefficient of resistance behaviour

#### **INTRODUCTION**

Nowadays, the generation of clean and efficient energy is becoming a global requirement. A wide range of materials like CeO<sub>2</sub> based fluorites, stabilized  $\delta$ -Bi<sub>2</sub>O<sub>3</sub>, yttria-stabilized zirconia (YSZ), perovskites, intergrowth perovskite-Bi<sub>2</sub>O<sub>3</sub> layers, pyrochlore etc have been studied as the oxygen ion conductors. In recent years, the structural and electrical properties of double perovskite-type oxides with the chemical formula A<sub>2</sub>B'B"O<sub>6</sub> have been studied by various researchers due to their various advanced technological applications. Rare earth cation based double perovskites (RE<sub>2</sub>B'B''O<sub>6</sub>, RE = rare earth, B'/B'' = different transition metals) exhibit various structural and physical properties due to their compositional variations [1]. The structural, electrical and optical properties of Gd<sub>2</sub>NiMnO<sub>6</sub> have earlier been studied [2] and the Rietveld refined XRD pattern confirmed the single-phase monoclinic structure with space group P2<sub>1</sub>/n with the distribution of ions at crystallographic Wyckoff positions 4*e* for Gd<sup>3+</sup> ions, 2*c* for Ni<sup>2+</sup> ions, 2*d* for Mn<sup>4+</sup> ions, and 4*e* for O<sup>2-</sup> ions. Each Ni<sup>2+</sup> and Mn<sup>4+</sup> ions surrounded by six O<sup>2-</sup> ions constitute the NiO<sub>6</sub> and MnO<sub>6</sub> octahedra, respectively. M. Rudra et al. [3] reported the temperature-dependent conductivity mechanisms observed in Pr<sub>2</sub>NiTiO<sub>6</sub>. They also reported that the conductivity, impedance and relaxation spectra highlight the combined contributions of the electrodes, grain-boundaries and grain microstructures to the carrier dynamics in Pr<sub>2</sub>NiTiO<sub>6</sub>.

This work is focused on the synthesis, structural and electrical properties of double perovskites  $Gd_2NiTiO_6$  which is synthesized by the sol-gel combustion method. The structure of the material was characterized using XRD and the electrical properties were studied using ac conductivity spectra.

#### EXPERIMENTAL

#### **Materials and Methods**

Gd<sub>2</sub>NiTiO<sub>6</sub> (GNT) double perovskite was synthesized by the solution combustion method. Gd<sub>2</sub>O<sub>3</sub> (Sigma-Aldrich, 99.9%), Nickel (II) nitrate hexahydrate [Ni(NO<sub>3</sub>)<sub>2</sub>.6H<sub>2</sub>O] (Merck, 98%), Tetra butyl titanate (TNBT) [C<sub>6</sub>H<sub>36</sub>O<sub>4</sub>Ti] (SRL) were used as the raw materials. At first, the stoichiometric amount of Gd<sub>2</sub>O<sub>3</sub> was dissolved in deionised water in a beaker and stirred using a magnetic stirrer. Nitric acid was mixed into this solution to obtain a clear solution. Next, a stoichiometric amount of Ni(NO<sub>3</sub>)<sub>2</sub>.6H<sub>2</sub>O and TNBT were mixed to the clear solution. Citric acid was mixed to this solution to the molar ratio of (C<sub>6</sub>H<sub>8</sub>O<sub>7</sub>): (Gd, Ni, Ti) = 3:1. The total solution was heated and stirred at 80°C for 8 h to get a homogeneous mixture that becomes gel due to evaporation of water. The gel was dried and partially auto ignited. To complete the ignition process, the partially ignited sample was transferred into a furnace and heated at 200°C for 2 h. The ignited sample was then collected and ground in an agate mortar. The as-prepared powder was calcined at 600°C for 2 h. Part of the powder was pressed to obtain a cylindrical pellet with a diameter of 10 mm. The pellets and powder samples were sintered in air at 1200°C for 4 h and used for various analyses.

For structural investigation of the sintered material X-Ray diffraction (XRD) data was taken by X-ray diffractometer (BRUKER, Model D8 Advance–AXS) using CuK $\alpha$  radiation [ $\lambda$  = 1.5406 Å] from 2 $\theta$  = 20° to 80° with step size 0.02°. Structural parameters were obtained for the samples using the Rietveld refinement method. For electrical measurement, the pellets were polished to acquire a smooth surface, and a high-temperature conductive graphite pest (Merck) was applied on both sides of the pellets to make the electrodes. Electrical measurements were carried out using two probe methods inside a tube furnace. An LCR meter (HIOKI Model: 3532-50) interfaced with a PC was used to collect the electrical data in the frequency ranging from 42 Hz to 5 MHz and in the temperature range from room temperature to 300 °C.

# Serial No. 462 isbnhelp[at]inflibnet[dot]ac[dot]in

♦ Skip to main content
♦ Skip to Navigation
♦ Screen Reliar

**ISBN** Allotted 1867464

Text Size: A- A A+



### Search ISBN

|              | 🗹 Advance Search |              |
|--------------|------------------|--------------|
| Book Title   | Email            | Name of Autł |
| Name of Pub  | Year             | 978-81-9553  |
| Select Produ | Select Langu     |              |
Search

Export to Excel

Search:

| #      | Book Title  | ISBN                  | Product<br>Form                   | Language        |
|--------|---|-----------------------|-----------------------------------|-----------------|
| 1      | Proceedings of the National<br>Conference on Materials, Devices<br>and Circuits in Communication<br>Technology, 2020 (MDCCT 2020)<br>(Edition -1) | 978-81-955389-<br>3-5 | Digital<br>download and<br>online | English         |
| Showir | ng 1 to 1 of 1 entries  |                       |                                   | Previous 1 Next |
| •      |   |                       |                                   | •               |



AddressFor General QueryRaja Rammohun Roy ISBN Agency<br/>Department of Higher Education,<br/>Ministry of Education<br/>Room No. 13, Jeevan Deep Building,<br/>4th Floor, Parliament Street,<br/>New Delhi -110001, India.Image: Constant of the street of the st

For Technical Query

**Important Links** 

- **L** +91-79-2326 8229
- FAQs
- Contact Us

Copyright © 2023 Website Managed and Maintained by INFLIBNET CENTRE, INFOCITY, GANDHINAGAR-382 007, GUJARAT, INDIA.

# Synthesis, characterisation, and electrical property study of ZnSnO<sub>3</sub> nanopowder

Mamotaj Khatun, Bithika Mandal, Sanjoy Mukherjee and Partha Mitra Dept. of Physics, The University of Burdwan, Burdwan - 713104, India Email: mamotajkhatunwb@gmail.com & mitrapartha1@rediffmail.com

#### Abstract

This work primarily focuses on grain growth dependent electrical properties of orthorhombic ZnSnO<sub>3</sub> prepared by co-precipitation method. Effects of sintering temperature (from 500°C to 650°C) on the size of grain, surface morphology and electrical properties were investigated. Single phase and crystallinity improvement was confirmed by XRD spectrum. Crystallite size of ZnSnO<sub>3</sub> powders were observed from ~9.02 nm to ~12.23 nm, calculated from XRD patterns using Scherrer equation. Grain growths of the samples were observed from the FESEM micrographs which are in good agreement with the crystallite size calculated from Scherrer equation. Complex impedance spectroscopy (CIS) analysis shows non-Debye type depressed semi-circular nature of the Nyquist plot. Frequency variable conductivity curves of the prepared samples show conductivity value increases with increasing temperature as well as increasing sintering temperature. Thus the mechanism confirms thermally activated process. Conductivity was higher for sample sintered at higher temperature (650°C) which might be associated with enhancement of mobility of carriers.

#### **INTRODUCTION**

In recent years, many researchers have attracted towards nanosized perovskite metal oxide compounds due to their widespread and praiseworthy application prospects in different fields. These n-type semiconducting perovskites possess unique properties due to their high surface to volume ratio, quantum confinement effect, better reaction rate etc. [1]. Perovskite oxides have the general formula ABO<sub>3</sub> where A- cation is larger in size than B- cation. The B-cation is in 6-fold co-ordination whereas the A- cation is in 12-fold cuboctahedral coordination in the crystal structure. Among different nanostructured perovskites, ZnSnO<sub>3</sub> exhibits interesting application in various electrical, electrochemical, optoelectronic and magneto electronic devices like solid oxide fuel cells (SOFCs), energy conversion devices, lithium-ion batteries, super capacitors, gas sensors, oxygen sensors, photocatalyst etc. In addition, presence of edge sharing SnO6-ZnO6 octahedral framework in orthorhombic crystal system of ZnSnO<sub>3</sub> ensures dispersed conduction band mediated enhanced mobility of charge carriers exerts remarkable role in the electrical properties [2]. Several processing routes such as solid state calcinations, spray pyrolysis, hydrothermal method, chemical method, and mechano-chemical milling method have been employed to synthesize Zinc stannate nano-particles [3]. The formation of ZnSnO3 was affected by the reaction conditions such as molar ratio of the precursors, reaction temperature, reaction time, pH of the solution, reaction rate etc. In the present work, we have made an attempt to synthesize well crystalline  $ZnSnO_3$  samples by simple chemical co-precipitation method. The effect of sintering temperature on the crystallite size and physicochemical properties of the ZnSnO<sub>3</sub> sample has been studied. To investigate the crystalline phase and morphology of the samples, XRD and FESEM were employed. The basic motivation for the present work was to study the influence of grain size on electrical properties. Impedance and conductivity study of the fabricated samples establishes structure-property correlation quite well.

#### **EXPERIMENTS**

#### A. Material Synthesis

In this paper, nanosized  $ZnSnO_3$  particles were synthesized using a novel co-precipitation method, which produces good crystalline materials without any further secondary phase. The preparation of  $ZnSnO_3$  was based on the following reaction:

$$ZnAc_2 + SnCl_4 + 6NaOH = ZnSnO_3 + 4NaOH + 2NaAc + 3H_2O$$
 (1)

0.5 M zinc acetate (ZnAc<sub>2</sub>.2H<sub>2</sub>O, analytical grade) and 0.5 M tin tetrachloride (SnCl<sub>4</sub>. 5H<sub>2</sub>O, analytical grade) were used as precursors without further purification, and dissolved into deionized water to form two separate transparent solutions respectively. The two solutions were mixed together. Then sodium hydroxide (NaOH) solution was added dropwise into the mixture under magnetic stirring until the pH of solution ~10. The white precipitate was collected, filtered, washed with deionized water, and dried at 100 °C to eliminate the unwanted soluble ions. The as- synthesized powder was initially hand grinded with a pestle and mortar. Then it was

# Serial No. 463 **SPRINGER NATURE** Link

**Ξ** Menu

**Q** Search

Home > New Horizons in Millimeter-Wave, Infrared and Terahertz Technologies > Chapter

## An Alternative Scheme of Quantum Optical Superfast Tristate CNOT Gate Using Frequency **Encoding Principle of Light with Semiconductor Optical Amplifier**

Chapter First Online: 01 November 2022

| pp 187–195 | Cite this chapter

New Horizons in Millimeter-Wave, Infrared and Terahertz **Technologies** 

Snigdha Hazra 🏹 & Sourangshu Mukhopadhyay

**Part of the book series:** Lecture Notes in Electrical Engineering ((LNEE, volume 953))

ි 248 Accesses

### Abstract

Photon is established as a strong and promising candidate in all-optical signal processing and superfast computing. Again, it is found that photon has a successful approach to be used as a quantum mechanical particle. For this reason, it is used as a carrier of information in optical systems instead of electrons in electronic systems. All-optical

🕁 Cart



### Serial No. 463 <u>M\_based\_on\_wavelength\_conversion\_using\_semiconductor\_optical\_amplifier\_and\_</u> filter

- 24. Mukhopadhyay S, Dey S, Saha S (2019) Photonics: a dream of modern technology. In: Photonics and fiber optics. CRC Press, pp. 39–66. https://doi.org/10.1201/9780429026584-2
- **25.** Dutta S, Mukhopadhyay S (2010) An all optical approach of frequency encoded NOT based Latch using semiconductor optical amplifier. J Opt 39:39–45

Article Google Scholar

## Acknowledgements

The authors acknowledge the financial support received from the DST-Inspire for extending a research fellowship to Snigdha Hazra.

## **Author information**

### **Authors and Affiliations**

Department of Physics, The University of Burdwan, Burdwan, West Bengal, 713104, India Snigdha Hazra & Sourangshu Mukhopadhyay

### **Corresponding author**

Correspondence to Snigdha Hazra.

## **Editor information**

## **Editors and Affiliations**

Electronics and Communication Engineering, Cooch Behar Government Engineering College, Cooch Behar, India Aritra Acharyya

School of Mines and Metallurgy, Kazi Nazrul University, Asansol, India

Arindam Biswas

Research Institute of Electronics, Shizuoka University, Hamamatsu, Japan

Hiroshi Inokawa

### **Rights and permissions**

**Reprints and permissions** 

## **Copyright information**

© 2022 The Author(s), under exclusive license to Springer Nature Singapore Pte Ltd.

## About this chapter

### Cite this chapter

Hazra, S., Mukhopadhyay, S. (2022). An Alternative Scheme of Quantum Optical Superfast Tristate CNOT Gate Using Frequency Encoding Principle of Light with Semiconductor Optical Amplifier. In: Acharyya, A., Biswas, A., Inokawa, H. (eds) New Horizons in Millimeter-Wave, Infrared and Terahertz Technologies. Lecture Notes in Electrical Engineering, vol 953. Springer, Singapore. https://doi.org/10.1007/978-981-19-6301-8\_16

.RIS ± .ENW ± .BIB ±

DOI https://doi.org/10.1007/9 78-981-19-6301-8\_16

Published 01 November 2022 Publisher Name Springer, Singapore

Print ISBN 978-981-19-6300-1

Online ISBN 978-981-19-6301-8 eBook Packages Physics and Astronomy Physics and Astronomy (R0)

# Serial No. 464 SPRINGER NATURE Link

 $\equiv$  Menu

**Q** Search

Home > New Horizons in Millimeter-Wave, Infrared and Terahertz Technologies > Chapter

## Use of Frequency Encoding Principle for Implementing Nano-Photonic Ultrafast Tristate Pauli X Gate

| Chapter | First Online: 01 November 2022

| pp197–204 | <u>Cite this chapter</u>



Ayan Dey 🔽 & Sourangshu Mukhopadhyay

Part of the book series: Lecture Notes in Electrical Engineering ((LNEE, volume 953))

**3230** Accesses **1** Citations

## Abstract

Optics has been found significant to take part in Quantum Computation as a carrier signal. In the last few years number of research articles were published where optics is used massively for implementing Quantum gates. In this chapter, an all-optical tristate Pauli X Gate has been designed with SOA-based optical switches using Frequency-encoded principles. Here, this design has three input signals and three output channels as

ᆬ Cart



**28.** Garai SK (2010) A scheme of developing frequency encoded tristate-optical logic operation using semiconductor optical amplifier. J Modern Optics 57(6)

**Google Scholar** 

**Google Scholar** 

**29.** Nakamura S, Tajima K, Sugimoto Y (1994) 10 ps all-optical switching in novel Mach–Zehnder configuration based on band-filling nonlinearity of GaAs. In: Conference on Lasers and Electro-Optics (CELO'94), CThS2

#### **Google Scholar**

**30.** Tajima K (1993) All-optical switch with switch-off time unrestricted by carrier lifetime. Jpn J Appl Phys 32:L1746–L1749

Article CAS Google Scholar

### Acknowledgements

The authors acknowledge the UGC-CAS proposal of the Department of Physics, The University of Burdwan, for some infrastructural support.

## **Author information**

### **Authors and Affiliations**

Department of Physics, The University of Burdwan, Golapbag, Burdwan, West Bengal, 713104, India Ayan Dey & Sourangshu Mukhopadhyay

### **Corresponding author**

Correspondence to Ayan Dey.

## Serial No. 464 Editor information

## **Editors and Affiliations**

Electronics and Communication Engineering, Cooch Behar Government Engineering College, Cooch Behar, India Aritra Acharyya

School of Mines and Metallurgy, Kazi Nazrul University, Asansol, India Arindam Biswas

Research Institute of Electronics, Shizuoka University, Hamamatsu, Japan Hiroshi Inokawa

# **Rights and permissions**

Reprints and permissions

# **Copyright information**

 $^{\odot}$  2022 The Author(s), under exclusive license to Springer Nature Singapore Pte Ltd.

# About this chapter

## Cite this chapter

Dey, A., Mukhopadhyay, S. (2022). Use of Frequency Encoding Principle for Implementing Nano-Photonic Ultrafast Tristate Pauli X Gate. In: Acharyya, A., Biswas, A., Inokawa, H. (eds) New Horizons in Millimeter-Wave, Infrared and Terahertz Technologies. Lecture Notes in Electrical Engineering, vol 953. Springer, Singapore. https://doi.org/10.1007/978-981-19-6301-8\_17

<u>.RIS</u>坐 <u>.ENW</u>坐 <u>.BIB</u>坐

Serial No. 464 DOI

https://doi.org/10.1007/9 78-981-19-6301-8\_17 Published 01 November 2022 Publisher Name Springer, Singapore

Print ISBN 978-981-19-6300-1 Online ISBN 978-981-19-6301-8 eBook Packages Physics and Astronomy Physics and Astronomy (R0)

### **Publish with us**

Policies and ethics [7]

# Serial No. 465 SPRINGER NATURE Link

 $\equiv$  Menu

**Q** Search

🔆 Cart

Home > Data Science and Communication > Conference paper

## Planar Branch-Line Coupler for 5G Communication

| Conference paper | First Online: 03 January 2024

| pp 763–771 | Cite this conference paper



**Data Science and Communication** 

(ICTDsC 2023)

Susmita Samanta, Joydeep Chakravorty, Anindya Bose & Tanmoy Sarkar 🖂

Part of the book series: <u>Studies in Autonomic, Data-driven and Industrial Computing</u> ((SADIC))

Included in the following conference series: International Conference on Data Science and Communication

**D** 219 Accesses

## Abstract

One of the essential components of many microwave circuits is the branch-line coupler (BLC). In this work, a BLC using apertures and shorting post is investigated to obtain a broad-band response characteristic. A delay line has also been introduced to maintain its

Login

**Department of Physics, The University of Burdwan, Burdwan, 713104, India** Susmita Samanta, Joydeep Chakravorty, Anindya Bose & Tanmoy Sarkar

## **Corresponding author**

Correspondence to <u>Tanmoy Sarkar</u>. **Editor information** 

### **Editors and Affiliations**

**Faculty of Engineering, University of Porto (FEUP), Porto, Portugal** João Manuel R. S. Tavares

National Institute of Telecommunications (INATEL), Santa Rita do Sapucaí, Minas Gerais, Brazil Joel J. P. C. Rodrigues

**Siliguri Institute of Technology, Siliguri, West Bengal, India** Debajyoti Misra

**Siliguri Institute of Technology, Siliguri, West Bengal, India** Debasmriti Bhattacherjee

## **Rights and permissions**

**Reprints and permissions** 

## **Copyright information**

© 2024 The Author(s), under exclusive license to Springer Nature Singapore Pte Ltd.

## About this paper

## Cite this paper

Samanta, S., Chakravorty, J., Bose, A., Sarkar, T. (2024). Planar Branch-Line Coupler for 5G Communication. In: Tavares, J.M.R.S., Rodrigues, J.J.P.C., Misra, D., Bhattacherjee, D. (eds)

Data Science and Communication. ICTDsC 2023. Studies in Autonomic, Data-driven and Industrial Computing. Springer, Singapore. https://doi.org/10.1007/978-981-99-5435-3\_56

.RIS ± .ENW ± .BIB ±

| DOI                       | Published         | Publisher Name           |
|---------------------------|-------------------|--------------------------|
| https://doi.org/10.1007/9 | 03 January 2024   | Springer, Singapore      |
| 78-981-99-5435-3_56       |                   |                          |
|                           |                   |                          |
| Print ISBN                | Online ISBN       | eBook Packages           |
| 978-981-99-5434-6         | 978-981-99-5435-3 | Intelligent Technologies |
|                           |                   | and Robotics             |
|                           |                   | Intelligent Technologies |
|                           |                   | and Robotics (R0)        |
|                           |                   |                          |

## **Publish with us**

Policies and ethics [7

# Serial No. 466 isbnhelp[at]inflibnet[dot]ac[dot]in

♦ Skip to main content
♦ Skip to Navigation
♦ Screen Reliar

Text Size: A- A A+



Search ISBN

|              | Advance Search |              |
|--------------|----------------|--------------|
| Book Title   | Email          | Name of Autł |
| Name of Pub  | Year           | 978-93-9073  |
| Select Produ | Select Langu   |              |

**ISBN** Allotted 1867464

Export to Excel

Search: Product # **Book Title** ISBN Form Language 1 Recent Developments in 978-93-90734-Book English **Chemical Sciences** 37-5 Previous 1 Next Showing 1 to 1 of 1 entries 4 **Total Visitors** 5331247 **Address For General Query** isbn-mhrd[at]gov[dot]in Raja Rammohun Roy ISBN Agency Department of Higher Education, **L** +91-11-2334 1739 Ministry of Education Room No. 13, Jeevan Deep Building, 4th Floor, Parliament Street, New Delhi -110001, India.

For Technical Query Im

☑ isbnhelp[at]inflibnet[dot]ac[dot]in ► User Manual

- ► FAQs
- Contact Us

Copyright © 2023 Website Managed and Maintained by INFLIBNET CENTRE, INFOCITY, GANDHINAGAR-382 007, GUJARAT, INDIA.

#### BOOK OF THE MONTH:

REVOLUTIONIZING OPHTHALMOLOGY: THE INTEGRATION OF

ARTIFICIAL INTELLIGENCE ALGORITHMS



Menu

☆ Home > Shop > Imprints > Nova > The Future of Green Synthesis





# The Future of Green Synthesis

\$160.00 - \$252.00

Jacques Kent (Editor)

Series: Green Research, Developments and Programs BISAC: SCI013080 DOI: https://doi.org/10.52305/YIWU6826

Currently, the synthesis process is one that is associated with toxic, non-biodegradable chemicals that are not safe for our environment. This book focuses on the trends, recent advancements and future of green synthesis. It discusses the advantages green synthesis has in terms of creating a more environmentally friendly, non-toxic, and less time-consuming method of synthesis.

\*\*Order the printed version and SAVE 50% on the e-book with Print+eBook. Price indicated includes shipping\*\*

Binding Hardcover ✓ ♂ Clear Publication Date: May 2023

Status: AV Page Count: 230

\$160.00

1

Add to cart

ISBN: 979-8-88697-816-2

Categories: 2023, Books, Environmental Chemistry, Green Research, Developments and Programs, Nova, Science and Technology

Like 1

### Table of Contents

Preface

Chapter 1. Microwave Assisted Synthesis: A Green Chemistry Approach and Future Directions S. Makwana and P. Kumari

Department of Chemistry, Sardar Vallabhbhai National Institute of Technology, Gujarat, India

# Chapter 2. Plant Extract-Based Green Synthesis of Zinc Oxide Nanomaterials and Its Greener Prospects for the Future

Subhodeep Barman<sup>1</sup>,<sup>2</sup> and Rahul Das<sup>1</sup>

<sup>1</sup>Department of Physics, The University of Burdwan, Golapbag, West Bengal, India <sup>2</sup>Department of Physics, Gangarampur College, Dakshin Dinajpur, West Bengal, India

#### Chapter 3. Biomolecules in Plant Extracts as Reducing Agents for AgNPs Synthesis

Yared Gutiérrez-Pinzón<sup>1</sup>, MD, Gilberto Velázquez-Juárez<sup>1</sup>, PhD, Edgar José López-Naranjo<sup>2</sup>, PhD and Diego Alberto Lomelí-Rosales<sup>1</sup>, PhD

<sup>1</sup>Departamento de Química, Centro Universitario de Ciencias Exactas e Ingenierías, Universidad de Guadalajara, Guadalajara, Jalisco, México

<sup>2</sup>Departamento de Ingeniería de Proyectos, Universidad de Guadalajara, Zapopan, Jalisco, México

#### Chapter 4. Trends and Recent Advancements in the Application of Microbial Enzymes in Green Chemistry, Biotransformation and Sustainable Development Amit Seth

Department of Life Sciences (Botany), Manipur University, Canchipur, Imphal, India

Chapter 5. Green Synthesis of Azole Derivatives as Antifungal Agents and Their Biomedical Applications

<sup>1</sup>Department of Chemistry, Veer Surendra Sai University of Technology, Burla, Sambalpur, Odisha, India

<sup>2</sup>Department of 5-yr Integrated Chemistry, Palamuru University Mahabubnagar, Telangana, India

# Chapter 6. Synthetic Possibilities for Hydroxy-Methylenebisphosphonic Acids and Their Derivatives

Zsuzsanna Szalai

Department of Organic Chemistry and Technology, Faculty of Chemical Technology and Biotechnology, Budapest University of Technology and Economics, Budapest, Hungary

**Chapter 7. The Exploration of Greener Technologies in the Synthesis of Anti-COVID-19 Drugs** Trinath Biswal<sup>1</sup>, D. S. Ramakrishna<sup>1</sup> and K. Sridhar Reddy<sup>2</sup>

<sup>1</sup>Dept. of Chemistry, Veer Surendra Sai University of Technology, Burla, Sambalpur, Odisha, India

<sup>2</sup>Dept. of Chemistry, Palamuru University Mahabubnagar, Telangana, India

#### Index

### Nova Science Publishers, Inc.

Nova publishes a wide array of books and journals from authors around the globe, focusing on Medicine and Health, Science and Technology and the Social Sciences and Humanities.

We publish over 1,500 new titles per year by leading researchers each year, and have a network of expert authors, editors and advisors spanning the global academic community in pursuit of advanced research developments.

Headquarters: Nova Science Publishers, Inc. 415 Oser Avenue, Suite N Hauppauge, NY, 11788 USA

Tel: 1-631-231-7269 Fax: 1-631-231-8175

#### Serial No. 469 MaterialsResearchForum LLC Publisher in Materials Science and Engineering

<section-header><section-header><section-header><section-header><text><text>

Home / Chapters / Future Directions in Nanomaterials Research for Biological Applications

1

Future Directions in Nanomaterials Research for Biological Applications \$30.00

Add to cart

Category: Chapters Tags: Biological Applications, Diagnostic Tools, Drug Delivery, Nanofabrications, Nanomaterials, Nanorobots, Tissue Engineering

#### Description

## Future Directions in Nanomaterials Research for Biological Applications

Rahul Das, Manab Deb Adhikari, Pratap Singh Chauhan

Nowadays, nanomaterial-based technologies have reached a great height from the application point of view. However, this chapter is mainly focused on nanomaterials research for biological applications and its future directions. Depending on the shape, size and elemental composition, nanomaterials are capable of exhibiting some unique and remarkable functional properties. Due to such functional properties, nanomaterials have attracted much attention for biomedical applications and are being tested for easier treatment and

diagnosis without any side effects. This chapter not only provides the expected biological applications of nanomaterials but also points out some information starting from key properties of biologically applicable nanomaterials to characterization through fabrication processes.

#### Keywords

Nanomaterials, Nanofabrications, Biological Applications, Diagnostic Tools, Drug Delivery, Tissue Engineering, Nanorobots

Published online 2/10/2024, 26 pages

**Citation:** Rahul Das, Manab Deb Adhikari, Pratap Singh Chauhan, Future Directions in Nanomaterials Research for Biological Applications, Materials Research Foundations, Vol. 161, pp 1-26, 2024

DOI: https://doi.org/10.21741/9781644902998-1

#### Part of the book on Nanoparticle Toxicity and Compatibility

#### References

[1] A.P. Ramos, M.A.E. Cruz, C.B. Tovani, P. Ciancaglini, Biomedical applications of nanotechnology, Biophys. Rev. 9 (2017) 79–89.

https://doi.org/10.1007/s12551-016-0246-2.

[2] E. Boisselier, D. Astruc, Gold nanoparticles in nanomedicine: preparations, imaging, diagnostics, therapies and toxicity, Chem. Soc. Rev. 38 (2009) 1759–1782. https://doi.org/10.1039/b806051g

[3] Q.A. Pankhurst, N.K.T. Thanh, S.K. Jones, J. Dobson, Progress in applications of magnetic nanoparticles in biomedicine, J. Phys. D. Appl. Phys. 42 (2009) 224001. https://doi.org/10.1088/0022-3727/42/22/224001

[4] T. Jamieson, R. Bakhshi, D. Petrova, R. Pocock, M. Imani, A.M. Seifalian, Biological applications of quantum dots, Biomaterials. 28 (2007) 4717–4732. https://doi.org/10.1016/j.biomaterials.2007.07.014

[5] Q.A. Pankhurst, J. Connolly, S.K. Jones, J. Dobson, Applications of magnetic nanoparticles in biomedicine, J. Phys. D. Appl. Phys. 36 (2003) R167–R181. https://doi.org/10.1088/0022-3727/36/13/201

[6] R. Kramer, D. Cohen, Functional genomics to new drug targets, Nat. Rev. Drug Discov. 3 (2004) 965–972. https://doi.org/10.1038/nrd1552.

[7] Y. Chen, H. Cong, Y. Shen, B. Yu, Biomedical application of manganese dioxide nanomaterials, Nanotechnology. 31 (2020) 202001.

https://doi.org/10.1088/1361-6528/ab6fe1

[8] F.S. Alves, J.N. Cruz, I.N. de Farias Ramos, D.L. do Nascimento Brandão, R.N. Queiroz, G.V. da Silva, G.V. da Silva, M.F. Dolabela, M.L. da Costa, A.S. Khayat, J. de Arimatéia Rodrigues do Rego, D. do Socorro Barros Brasil, Evaluation of Antimicrobial Activity and Cytotoxicity Effects of Extracts of Piper nigrum L. and Piperine, Separations. 10 (2023) 21.

https://doi.org/10.3390/separations10010021

[9] P. Podsiadlo, S. Paternel, J.M. Rouillard, Z. Zhang, J. Lee, J.W. Lee, E. Gulari, N.A. Kotov, Layer-by-layer assembly of nacre-like nanostructured composites with antimicrobial properties, AIChE Annu. Meet. Conf. Proc. 21 (2005) 5198-5210. https://doi.org/10.1021/la051284+ [10] S. Murthy, P. Effiong, C.C. Fei, Metal oxide nanoparticles in biomedical applications, Met. Oxide Powder Technol. Fundam. Process. Methods Appl. (2020) 233-251. https://doi.org/10.1016/B978-0-12-817505-7.00011-7. [11] D. Ung, L.D. Tung, G. Caruntu, D. Delaportas, I. Alexandrou, I.A. Prior, N.T.K. Thanh, Variant shape growth of nanoparticles of metallic Fe-Pt, Fe-Pd and Fe-Pt-Pd alloys, CrystEngComm. 11 (2009) 1309–1316. https://doi.org/10.1039/b823290n [12] T. Jamieson, R. Bakhshi, D. Petrova, R. Pocock, M. Imani, A.M. Seifalian, Biological applications of quantum dots, Biomaterials. 28 (2007) 4717–4732. https://doi.org/10.1016/j.biomaterials.2007.07.014 [13] J. Drbohlavova, V. Adam, R. Kizek, J. Hubalek, Quantum dots – characterization, preparation and usage in biological systems, Int. J. Mol. Sci. 10 (2009) 656-673. https://doi.org/10.3390/ijms10020656 [14] A.L. Rogach, A. Eychmüller, S.G. Hickey, S. V. Kershaw, Infrared-emitting colloidal nanocrystals: Synthesis, assembly, spectroscopy, and applications, Small. 3 (2007) 536-557. https://doi.org/10.1002/smll.200600625 [15] K. Shahane, M. Kshirsagar, S. Tambe, D. Jain, S. Rout, M.K.M. Ferreira, S. Mali, P. Amin, P.P. Srivastav, J. Cruz, R.R. Lima, An Updated Review on the Multifaceted Therapeutic Potential of Calendula officinalis L., Pharmaceuticals. 16 (2023) 611. https://doi.org/10.3390/ph16040611 [16] A.G. Roca, R. Costo, A.F. Rebolledo, S. Veintemillas-Verdaguer, P. Tartaj, T. González-Carreño, M.P. Morales, C.J. Serna, Progress in the preparation of magnetic nanoparticles for applications in biomedicine, J. Phys. D. Appl. Phys. 42 (2009) 224002. https://doi.org/10.1088/0022-3727/42/22/224002 [17] N.T.K. Thanh, L.A.W. Green, Functionalisation of nanoparticles for biomedical applications, Nano Today. 5 (2010) 213–230. https://doi.org/10.1016/j.nantod.2010.05.003 [18] Scientific and Clinical Applications of Magnetic Carriers, Sci. Clin. Appl. Magn. Carriers. (1997). https://doi.org/10.1007/978-1-4757-6482-6 [19] M. Chen, D.E. Nikles, Synthesis of spherical FePd and CoPt nanoparticles, J. Appl. Phys. 91 (2002) 8477-8479. https://doi.org/10.1063/1.1456406 [20] R. Shanmuganathan, I. Karuppusamy, M. Saravanan, H. Muthukumar, K. Ponnuchamy, V.S. Ramkumar, A. Pugazhendhi, Synthesis of Silver Nanoparticles and their Biomedical Applications – A Comprehensive Review, Curr. Pharm. Des. 25 (2019) 2650-2660. https://doi.org/10.2174/1381612825666190708185506 [21] I. Rezić, Nanoparticles for Biomedical Application and Their Synthesis, Polymers (Basel). 14 (2022) 4961. https://doi.org/10.3390/polym14224961 [22] C.S.S.R. Kumar, F. Mohammad, Magnetic nanomaterials for hyperthermiabased therapy and controlled drug delivery, Adv. Drug Deliv. Rev. 63 (2011)

789–808. https://doi.org/10.1016/j.addr.2011.03.008 [23] A.M. Schrand, M.F. Rahman, S.M. Hussain, J.J. Schlager, D.A. Smith, A.F. Sved, Metal-based nanoparticles and their toxicity assessment, Wiley Interdiscip. Rev. Nanomedicine Nanobiotechnology. 2 (2010) 544–568. https://doi.org/10.1002/wnan.103 [24] M.I. Anik, N. Mahmud, A. Al Masud, M. Hasan, Gold nanoparticles (GNPs) in biomedical and clinical applications: A review, Nano Sel. 3 (2022) 792–828. https://doi.org/10.1002/nano.202100255 [25] L. S Jairam, A. Chandrashekar, T.N. Prabhu, S.B. Kotha, M.S. Girish, I.M. Devraj, M. Dhanya Shri, K. Prashantha, A review on biomedical and dental applications of cerium oxide nanoparticles — Unearthing the potential of this rare earth metal, J. Rare Earths. 41 (2023) 1645–1661. https://doi.org/10.1016/j.jre.2023.04.009 [26] S. Pantic, S.R. Skodric, Z. Loncar, I. Pantic, Zinc oxide nanoparticles: Potential novel applications in cellular physiology, pathology, neurosciences and cancer research, Rev. Adv. Mater. Sci. 58 (2019) 17-21. https://doi.org/10.1515/rams-2019-0002 [27] S. Jha, R. Rani, S. Singh, Biogenic Zinc Oxide Nanoparticles and Their Biomedical Applications: A Review, J. Inorg. Organomet. Polym. Mater. 33 (2023) 1437-1452. https://doi.org/10.1007/s10904-023-02550-x [28] A. Esmaeilnejad, P. Mahmoudi, A. Zamanian, M. Mozafari, Synthesis of titanium oxide nanotubes and their decoration by MnO nanoparticles for biomedical applications, Ceram. Int. 45 (2019) 19275–19282. https://doi.org/10.1016/j.ceramint.2019.06.177 [29] S. Jafari, B. Mahyad, H. Hashemzadeh, S. Janfaza, T. Gholikhani, L. Tayebi, Biomedical applications of TiO2 nanostructures: Recent advances, Int. J. Nanomedicine. 15 (2020) 3447-3470. https://doi.org/10.2147/IJN.S249441 [30] S. Khan, A.A. Ansari, A. Malik, A.A. Chaudhary, J.B. Syed, A.A. Khan, Preparation, characterizations and in vitro cytotoxic activity of nickel oxide nanoparticles on HT-29 and SW620 colon cancer cell lines, J. Trace Elem. Med. Biol. 52 (2019) 12–17. https://doi.org/10.1016/j.jtemb.2018.11.003 [31] N. Behera, M. Arakha, M. Priyadarshinee, B.S. Pattanayak, S. Soren, S. Jha, B.C. Mallick, Oxidative stress generated at nickel oxide nanoparticle interface results in bacterial membrane damage leading to cell death, RSC Adv. 9 (2019) 24888–24894. https://doi.org/10.1039/c9ra02082a [32] S. Naz, A. Gul, M. Zia, R. Javed, Synthesis, biomedical applications, and toxicity of CuO nanoparticles, Appl. Microbiol. Biotechnol. 107 (2023) 1039-1061. https://doi.org/10.1007/s00253-023-12364-z [33] P. Paramita, V.D. Subramaniam, R. Murugesan, M. Gopinath, I. Ramachandran, S. Ramalingam, X.F. Sun, A. Banerjee, F. Marotta, S. Pathak, Evaluation of potential anti-cancer activity of cationic liposomal nanoformulated Lycopodium clavatum in colon cancer cells, IET Nanobiotechnology. 12 (2018) 727-732. https://doi.org/10.1049/ietnbt.2017.0106

[34] T. Sabah, K.H. Jawad, N. Al-attar, Synthesis and Biomedical Activity of Aluminium Oxide Nanoparticles by Laser Ablation Technique, Res. J. Pharm. Technol. 16 (2023) 1267–1273. https://doi.org/10.52711/0974-360X.2023.00209 [35] S.M. Dizaj, F. Lotfipour, M. Barzegar-Jalali, M.H. Zarrintan, K. Adibkia, Antimicrobial activity of the metals and metal oxide nanoparticles, Mater. Sci. Eng. C. 44 (2014) 278-284. https://doi.org/10.1016/j.msec.2014.08.031 [36] M. Kumari, B. Sarkar, K. Mukherjee, Nanoscale calcium oxide and its biomedical applications: A comprehensive review, Biocatal. Agric. Biotechnol. 47 (2023) 102506. https://doi.org/10.1016/j.bcab.2022.102506 [37] S. Abinaya, H.P. Kavitha, Magnesium Oxide Nanoparticles: Effective Antilarvicidal and Antibacterial Agents, ACS Omega. 8 (2023) 5225–5233. https://doi.org/10.1021/acsomega.2c01450 [38] K. Szostak, P. Ostaszewski, J. Pulit-Prociak, M. Banach, Bismuth Oxide Nanoparticles in Drug Delivery Systems, Pharm. Chem. J. 53 (2019) 48-51. https://doi.org/10.1007/s11094-019-01954-9 [39] M.A. Shahbazi, L. Faghfouri, M.P.A. Ferreira, P. Figueiredo, H. Maleki, F. Sefat, J. Hirvonen, H.A. Santos, The versatile biomedical applications of bismuth-based nanoparticles and composites: Therapeutic, diagnostic, biosensing, and regenerative properties, Chem. Soc. Rev. 49 (2020) 1253–1321. https://doi.org/10.1039/c9cs00283a [40] M. Wu, P. Hou, L. Dong, L. Cai, Z. Chen, M. Zhao, J. Li, Manganese dioxide nanosheets: From preparation to biomedical applications, Int. J. Nanomedicine. 14 (2019) 4781-4800. https://doi.org/10.2147/IJN.S207666 [41] D.F. Williams, A model for biocompatibility and its evaluation, J. Biomed. Eng. 11 (1989) 185-191. https://doi.org/10.1016/0141-5425(89)90138-6 [42] M.A. Dobrovolskaia, S.E. McNeil, Immunological properties of engineered nanomaterials, Nat. Nanotechnol. 2 (2007) 469-478. https://doi.org/10.1038/nnano.2007.223 [43] M. Huston, M. Debella, M. Dibella, A. Gupta, Green synthesis of nanomaterials, Nanomaterials. 11 (2021) 2130. https://doi.org/10.3390/nano11082130 [44] D. Bokov, A. Turki Jalil, S. Chupradit, W. Suksatan, M. Javed Ansari, I.H. Shewael, G.H. Valiev, E. Kianfar, Nanomaterial by Sol-Gel Method: Synthesis and Application, Adv. Mater. Sci. Eng. 2021 (2021) 1–21. https://doi.org/10.1155/2021/5102014 [45] C. Dhand, N. Dwivedi, X.J. Loh, A.N. Jie Ying, N.K. Verma, R.W. Beuerman, R. Lakshminarayanan, S. Ramakrishna, Methods and strategies for the synthesis of diverse nanoparticles and their applications: A comprehensive overview, RSC Adv. 5 (2015) 105003-105037. https://doi.org/10.1039/c5ra19388e [46] A.E. Danks, S.R. Hall, Z. Schnepp, The evolution of "sol-gel" chemistry as a technique for materials synthesis, Mater. Horizons. 3 (2016) 91–112. https://doi.org/10.1039/c5mh00260e [47] S. Barman, S. Sikdar, A. Biswas, B.K. Mandal, R. Das, Structural microanalysis of green synthesized AlxZn(1-x)O nanoparticles, Nano Express. 1

(2020) 20003. https://doi.org/10.1088/2632-959X/ab9f54 [48] Y. Huang, C.Y. Haw, Z. Zheng, J. Kang, J.C. Zheng, H.Q. Wang, Biosynthesis of Zinc Oxide Nanomaterials from Plant Extracts and Future Green Prospects: A Topical Review, Adv. Sustain. Syst. 5 (2021). https://doi.org/10.1002/adsu.202000266 [49] S. Barman, S. Sikdar, A. Biswas, A. Islam, R. Das, Green synthesis of MnxZn(1-x)O nanostructure using Azadirachta indica leaf extract and its microstructural and optical study, Phys. Scr. 97 (2022) 45002. https://doi.org/10.1088/1402-4896/ac520c [50] M.H. Sarfraz, M. Zubair, B. Aslam, A. Ashraf, M.H. Siddique, S. Hayat, J.N. Cruz, S. Muzammil, M. Khurshid, M.F. Sarfraz, A. Hashem, T.M. Dawoud, G.D. Avila-Quezada, E.F. Abd\_Allah, Comparative analysis of phyto-fabricated chitosan, copper oxide, and chitosan-based CuO nanoparticles: antibacterial potential against Acinetobacter baumannii isolates and anticancer activity against HepG2 cell lines, Front. Microbiol. 14 (2023) 1188743. https://doi.org/10.3389/fmicb.2023.1188743 [51] S. Barman, S. Sikdar, R. Das, A comprehensive study on ZrO2-ZnO nanocomposites synthesized by the plant-mediated green method, Phys. Scr. 98 (2023) 85947. https://doi.org/10.1088/1402-4896/ace857 [52] B. Shkodra-Pula, A. Vollrath, U.S. Schubert, S. Schubert, Polymer-based nanoparticles for biomedical applications, Front. Nanosci. 16 (2020) 233–252. https://doi.org/10.1016/B978-0-08-102828-5.00009-7 [53] M. Ovais, I. Ahmad, A.T. Khalil, S. Mukherjee, R. Javed, M. Ayaz, A. Raza, Z.K. Shinwari, Wound healing applications of biogenic colloidal silver and gold nanoparticles: recent trends and future prospects, Appl. Microbiol. Biotechnol. 102 (2018) 4305-4318. https://doi.org/10.1007/s00253-018-8939-z [54] A.A.A. Aljabali, Y. Akkam, M.S. Al Zoubi, K.M. Al-Batayneh, B. Al-Trad, O.A. Alrob, A.M. Alkilany, M. Benamara, D.J. Evans, Synthesis of gold nanoparticles using leaf extract of ziziphus zizyphus and their antimicrobial activity, Nanomaterials. 8 (2018) 174. https://doi.org/10.3390/nano8030174 [55] B.K. Mandal, R. Mandal, S. Sikdar, S. Sarma, A. Srinivasan, S.R. Chowdhury, B. Das, R. Das, Green synthesis of NiO nanoparticle using Punica granatum peel extract and its characterization for methyl orange degradation, Mater. Today Commun. 34 (2023) 105302. https://doi.org/10.1016/j.mtcomm.2022.105302 [56] S.A. Razack, A. Suresh, S. Sriram, G. Ramakrishnan, S. Sadanandham, M. Veerasamy, R.B. Nagalamadaka, R. Sahadevan, Green synthesis of iron oxide nanoparticles using Hibiscus rosa-sinensis for fortifying wheat biscuits, SN Appl. Sci. 2 (2020). https://doi.org/10.1007/s42452-020-2477-x [57] S.C. Mali, A. Dhaka, C.K. Githala, R. Trivedi, Green synthesis of copper nanoparticles using Celastrus paniculatus Willd. leaf extract and their photocatalytic and antifungal properties, Biotechnol. Reports. 27 (2020) e00518. https://doi.org/10.1016/j.btre.2020.e00518 [58] Hemlata, P.R. Meena, A.P. Singh, K.K. Tejavath, Biosynthesis of Silver

Nanoparticles Using Cucumis prophetarum Aqueous Leaf Extract and Their Antibacterial and Antiproliferative Activity against Cancer Cell Lines, ACS Omega. 5 (2020) 5520–5528. https://doi.org/10.1021/acsomega.0c00155 [59] M. Naseer, U. Aslam, B. Khalid, B. Chen, Green route to synthesize Zinc Oxide Nanoparticles using leaf extracts of Cassia fistula and Melia azadarach and their antibacterial potential, Sci. Rep. 10 (2020).

https://doi.org/10.1038/s41598-020-65949-3

[60] S.E. Sandler, B. Fellows, O. Thompson Mefford, Best Practices for
Characterization of Magnetic Nanoparticles for Biomedical Applications, Anal.
Chem. 91 (2019) 14159–14169. https://doi.org/10.1021/acs.analchem.9b03518
[61] I. Rezić, Nanoparticles for Biomedical Application and Their Synthesis,
Polymers (Basel). 14 (2022) 4961. https://doi.org/10.3390/polym14224961
[62] M. Mabrouk, D.B. Das, Z.A. Salem, H.H. Beherei, Nanomaterials for
biomedical applications: Production, characterisations, recent trends and
difficulties, Molecules. 26 (2021) 1077.

https://doi.org/10.3390/molecules26041077

[63] S. Muzammil, J. Neves Cruz, R. Mumtaz, I. Rasul, S. Hayat, M.A. Khan, A.M. Khan, M.U. Ijaz, R.R. Lima, M. Zubair, Effects of Drying Temperature and Solvents on In Vitro Diabetic Wound Healing Potential of Moringa oleifera Leaf Extracts, Molecules. 28 (2023) 710. https://doi.org/10.3390/molecules28020710 [64] Z. Cai, Z. Ye, X. Yang, Y. Chang, H. Wang, Y. Liu, A. Cao, Encapsulated enhanced green fluorescence protein in silica nanoparticle for cellular imaging, Nanoscale. 3 (2011) 1974-1976. https://doi.org/10.1039/c0nr00956c [65] B.K. Mandal, R. Mandal, D. Limbu, M.D. Adhikari, P.S. Chauhan, R. Das, Green synthesis of AgCl nanoparticles using Calotropis gigantea: Characterization and their enhanced antibacterial activities, Chem. Phys. Lett. 801 (2022) 139699. https://doi.org/10.1016/j.cplett.2022.139699 [66] R. Weissleder, Scaling down imaging: Molecular mapping of cancer in mice, Nat. Rev. Cancer. 2 (2002) 11-18. https://doi.org/10.1038/nrc701 [67] S. Flacke, S. Fischer, M.J. Scott, R.J. Fuhrhop, J.S. Allen, M. McLean, P. Winter, G.A. Sicard, P.J. Gaffney, S.A. Wickline, G.M. Lanza, Novel MRI contrast agent for molecular imaging of fibrin implications for detecting vulnerable plagues, Circulation. 104 (2001) 1280-1285.

https://doi.org/10.1161/hc3601.094303

[68] R.B.M. de Almeida, D.B. Barbosa, M.R. do Bomfim, J.A.O. Amparo, B.S.
Andrade, S.L. Costa, J.M. Campos, J.N. Cruz, C.B.R. Santos, F.H.A. Leite, M.B.
Botura, Identification of a Novel Dual Inhibitor of Acetylcholinesterase and
Butyrylcholinesterase: In Vitro and In Silico Studies, Pharmaceuticals. 16 (2023)
95. https://doi.org/10.3390/ph16010095

[69] N. Arndt, H.D.N. Tran, R. Zhang, Z.P. Xu, H.T. Ta, Different Approaches to Develop Nanosensors for Diagnosis of Diseases, Adv. Sci. 7 (2020). https://doi.org/10.1002/advs.202001476

[70] L. Helm, Optimization of gadolinium-based MRI contrast agents for high magnetic-field applications, Future Med. Chem. 2 (2010) 385–396.

https://doi.org/10.4155/fmc.09.174

[71] J.A. Park, J.J. Lee, J.C. Jung, D.Y. Yu, C. Oh, S. Ha, T.J. Kim, Y. Chang, Gd-DOTA conjugate of RGD as a potential tumor-targeting MRI contrast agent.,
Chembiochem. 9 (2008) 2811–2813. https://doi.org/10.1002/cbic.200800529
[72] W.L. Zhang, D.W. Yong, J. Huang, J.H. Yu, S.Y. Liu, M.X. Fan, Fabrication of polymer-gadolinium (III) complex nanomicelle from poly(ethylene glycol)-polysuccinimide conjugate and diethylenetriaminetetraacetic acid-gadolinium as magnetic resonance imaging contrast agents, J. Appl. Polym. Sci. 120 (2011) 2596–2605. https://doi.org/10.1002/app.33464

[73] L.S. Karfeld-Sulzer, E.A. Waters, N.E. Davis, T.J. Meade, A.E. Barron, Multivalent protein polymer mri contrast agents: Controlling relaxivity via modulation of amino acid sequence, Biomacromolecules. 11 (2010) 1429– 1436. https://doi.org/10.1021/bm901378a

[74] N. Kamaly, A.D. Miller, Paramagnetic liposome nanoparticles for cellular and tumour imaging, Int. J. Mol. Sci. 11 (2010) 1759–1776.

https://doi.org/10.3390/ijms11041759

[75] H. Bin Na, I.C. Song, T. Hyeon, Inorganic nanoparticles for MRI contrast agents, Adv. Mater. 21 (2009) 2133–2148.

https://doi.org/10.1002/adma.200802366

[76] Y.Z. Shao, L.Z. Liu, S.Q. Song, R.H. Cao, H. Liu, C.Y. Cui, X. Li, M.J. Bie, L. Li, A novel one-step synthesis of Gd3+-incorporated mesoporous SiO2

nanoparticles for use as an efficient MRI contrast agent, Contrast Media Mol. Imaging. 6 (2011) 110–118. https://doi.org/10.1002/cmmi.412

[77] L. Nie, F. Liu, P. Ma, X. Xiao, Applications of gold nanoparticles in optical biosensors, J. Biomed. Nanotechnol. 10 (2014) 2700–2721.

https://doi.org/10.1166/jbn.2014.1987

[78] H. Daraee, A. Eatemadi, E. Abbasi, S.F. Aval, M. Kouhi, A. Akbarzadeh, Application of gold nanoparticles in biomedical and drug delivery, Artif. Cells, Nanomedicine Biotechnol. 44 (2016) 410–422.

https://doi.org/10.3109/21691401.2014.955107

[79] L.E. Cole, R.D. Ross, J.M. Tilley, T. Vargo-Gogola, R.K. Roeder, Gold nanoparticles as contrast agents in X-ray imaging and computed tomography, Nanomedicine. 10 (2015) 321–341. https://doi.org/10.2217/nnm.14.171
[80] D. Lombardo, M.A. Kiselev, M.T. Caccamo, Smart Nanoparticles for Drug Delivery Application: Development of Versatile Nanocarrier Platforms in Biotechnology and Nanomedicine, J. Nanomater. 2019 (2019) 1–26. https://doi.org/10.1155/2019/3702518

[81] D. Bobo, K.J. Robinson, J. Islam, K.J. Thurecht, S.R. Corrie, Nanoparticle-Based Medicines: A Review of FDA-Approved Materials and Clinical Trials to Date, Pharm. Res. 33 (2016) 2373–2387. https://doi.org/10.1007/s11095-016-1958-5

[82] R. Qiao, C. Yang, M. Gao, Superparamagnetic iron oxide nanoparticles:From preparations to in vivo MRI applications, J. Mater. Chem. 19 (2009) 6274–6293. https://doi.org/10.1039/b902394a

[83] A.S. Teja, P.Y. Koh, Synthesis, properties, and applications of magnetic iron oxide nanoparticles, Prog. Cryst. Growth Charact. Mater. 55 (2009) 22-45. https://doi.org/10.1016/j.pcrysgrow.2008.08.003 [84] J. Mürbe, A. Rechtenbach, J. Töpfer, Synthesis and physical characterization of magnetite nanoparticles for biomedical applications, Mater. Chem. Phys. 110 (2008) 426-433. https://doi.org/10.1016/j.matchemphys.2008.02.037 [85] C. Sun, J.S.H. Lee, M. Zhang, Magnetic nanoparticles in MR imaging and drug delivery, Adv. Drug Deliv. Rev. 60 (2008) 1252-1265. https://doi.org/10.1016/j.addr.2008.03.018 [86] J. Hong, P. Gong, D. Xu, H. Sun, S. Yao, Synthesis and characterization of carboxyl-functionalized magnetic nanogel via "Green" photochemical method, J. Appl. Polym. Sci. 105 (2007) 1882–1887. https://doi.org/10.1002/app.25655 [87] A.K. Gupta, M. Gupta, Synthesis and surface engineering of iron oxide nanoparticles for biomedical applications, Biomaterials. 26 (2005) 3995–4021. https://doi.org/10.1016/j.biomaterials.2004.10.012 [88] J. Pan, Y. Wang, S.S. Feng, Formulation, characterization, and in vitro evaluation of quantum dots loaded in poly(Lactide)-Vitamin E TPGS nanoparticles for cellular and molecular imaging, Biotechnol. Bioeng. 101 (2008) 622-633. https://doi.org/10.1002/bit.21924 [89] L. Yang, H. Mao, Y. Andrew Wang, Z. Cao, X. Peng, X. Wang, H. Duan, C. Ni, Q. Yuan, G. Adams, M.Q. Smith, W.C. Wood, X. Gao, S. Nie, Single chain epidermal growth factor receptor antibody conjugated nanoparticles for in vivo tumor targeting and imaging, Small. 5 (2009) 235–243. https://doi.org/10.1002/smll.200800714 [90] B. Ballou, L. Ernst, A. Waggoner, Fluorescence Imaging of Tumors In Vivo, Curr. Med. Chem. 12 (2005) 795-805. https://doi.org/10.2174/0929867053507324 [91] B. Ballou, L.A. Ernst, S. Andreko, T. Harper, J.A.J. Fitzpatrick, A.S. Waggoner, M.P. Bruchez, Sentinel lymph node imaging using quantum dots in mouse tumor models, Bioconjug. Chem. 18 (2007) 389-396. https://doi.org/10.1021/bc060261j [92] P. Wunderbaldinger, L. Josephson, C. Bremer, A. Moore, R. Weissleder, Detection of lymph node metastases by contrast-enhanced MRI in an experimental model, Magn. Reson. Med. 47 (2002) 292–297. https://doi.org/10.1002/mrm.10068 [93] O. Rabin, J.M. Perez, J. Grimm, G. Wojtkiewicz, R. Weissleder, An X-ray computed tomography imaging agent based on long-circulating bismuth sulphide nanoparticles, Nat. Mater. 5 (2006) 118–122. https://doi.org/10.1038/nmat1571 [94] M.A. Hahn, A.K. Singh, P. Sharma, S.C. Brown, B.M. Moudgil, Nanoparticles as contrast agents for in-vivo bioimaging: Current status and future perspectives, Anal. Bioanal. Chem. 399 (2011) 3-27. https://doi.org/10.1007/s00216-010-4207-5

[95] D.B. Elrod, R. Partha, D. Danila, S.W. Casscells, J.L. Conyers, An iodinated liposomal computed tomographic contrast agent prepared from a dijodophosphatidylcholine lipid, Nanomedicine Nanotechnology, Biol. Med. 5 (2009) 42-45. https://doi.org/10.1016/j.nano.2008.06.007 [96] S. Kweon, H.J. Lee, W.J. Hyung, J. Suh, J.S. Lim, S.J. Lim, Liposomes coloaded with iopamidol/lipiodol as a res-targeted contrast agent for computed tomography imaging, Pharm. Res. 27 (2010) 1408–1415. https://doi.org/10.1007/s11095-010-0135-5 [97] J. Zheng, C. Allen, S. Serra, D. Vines, M. Charrong, D.A. Jaffray, Liposome contrast agent for CT-based detection and localization of neoplastic and inflammatory lesions in rabbits: Validation with FDG-PET and histology, Contrast Media Mol. Imaging. 5 (2010) 147-154. https://doi.org/10.1002/cmmi.378 [98] A. Chrastina, J.E. Schnitzer, Iodine-125 radiolabeling of silver nanoparticles for in vivo SPECT imaging, Int. J. Nanomedicine. 5 (2010) 653–659. https://doi.org/10.2147/IJN.S11677 [99] J.L. Van Herck, G.R.Y. De Meyer, W. Martinet, R.A. Salgado, B. Shivalkar, R. De Mondt, H. Van De Ven, A. Ludwig, P. Van Der Veken, L. Van Vaeck, H. Bult, A.G. Herman, C.J. Vrints, Multi-slice computed tomography with N1177 identifies ruptured atherosclerotic plaques in rabbits, Basic Res. Cardiol. 105 (2010) 51-59. https://doi.org/10.1007/s00395-009-0052-0 [100] K.L. Aillon, N. El-Gendy, C. Dennis, J.P. Norenberg, J. McDonald, C. Berkland, Iodinated NanoClusters as an inhaled computed tomography contrast agent for lung visualization, Mol. Pharm. 7 (2010) 1274–1282. https://doi.org/10.1021/mp1000718 [101] S. Zalipsky, Chemistry of polyethylene glycol conjugates with biologically active molecules, Adv. Drug Deliv. Rev. 16 (1995) 157-182. https://doi.org/10.1016/0169-409X(95)00023-Z [102] Z. Li, S. Tan, S. Li, Q. Shen, K. Wang, Cancer drug delivery in the nano era: An overview and perspectives (Review), Oncol. Rep. 38 (2017) 611-624. https://doi.org/10.3892/or.2017.5718 [103] S. Katsuki, T. Matoba, J.I. Koga, K. Nakano, K. Egashira, Anti-inflammatory Nanomedicine for Cardiovascular Disease, Front. Cardiovasc. Med. 4 (2017). https://doi.org/10.3389/fcvm.2017.00087 [104] N.A. Ochekpe, P.O. Olorunfemi, N.C. Ngwuluka, Nanotechnology and drug delivery part 1: Background and applications, Trop. J. Pharm. Res. 8 (2009) 265-274. https://doi.org/10.4314/tjpr.v8i3.44546 [105] S. Sim, N.K. Wong, Nanotechnology and its use in imaging and drug delivery (Review), Biomed. Reports. 14 (2021). https://doi.org/10.3892/br.2021.1418 [106] M. Fathi-Achachelouei, H. Knopf-Marques, C.E. Ribeiro da Silva, J. Barthès, E. Bat, A. Tezcaner, N.E. Vrana, Use of Nanoparticles in Tissue Engineering and Regenerative Medicine, Front. Bioeng. Biotechnol. 7 (2019). https://doi.org/10.3389/fbioe.2019.00113

[107] Y.L. Colson, M.W. Grinstaff, Biologically responsive polymeric nanoparticles for drug delivery, Adv. Mater. 24 (2012) 3878–3886. https://doi.org/10.1002/adma.201200420
[108] M.P. Ajith, M. Aswathi, E. Priyadarshini, P. Rajamani, Recent innovations of nanotechnology in water treatment: A comprehensive review, Bioresour.
Technol. 342 (2021) 126000. https://doi.org/10.1016/j.biortech.2021.126000
[109] M. Rai, J.C. Dos Santos, M.F. Soler, P.R. Franco Marcelino, L.P. Brumano, A.P. Ingle, S. Gaikwad, A. Gade, S.S. Da Silva, Strategic role of nanotechnology for production of bioethanol and biodiesel, Nanotechnol. Rev. 5 (2016) 231– 250. https://doi.org/10.1515/ntrev-2015-0069
[110] A.A.G. Requicha, Nanorobots, NEMS, and nanoassembly, Proc. IEEE. 91
(2003) 1922–1933. https://doi.org/10.1109/JPROC.2003.818333
[111] S.M. Asil, J. Ahlawat, G.G. Barroso, M. Narayan, Application of nanotechnology in stem-cell-based therapy of neurodegenerative diseases,

Appl. Sci. 10 (2020) 4852. https://doi.org/10.3390/app10144852

#### Bookstore search

Search products... Search

#### Recent Books

**Emerging Materials for Next Frontier Energy and Environment Applications** \$95.00

**Green Synthesis and Emerging Applications of Frontier Nanomaterials** \$125.00

#### **Terotechnology XIII** \$0.00 – \$125.00

Materials Research - 2024 Fall Edition Catalog \$0.00

**The Strength-Ductility Paradox** \$95.00

**Metal Forming 2024** \$0.00 - \$150.00

Superhydrophobic Metal Surfaces \$95.00

Artificial Intelligence and Alloy Design

















**Renewable Energy: Generation and Application - ICREGA'24** \$0.00 - \$125.00

Aerospace Science and Engineering - IV Aerospace PhD-Days \$0.00 - \$95.00



© Materials Research Forum. All Rights Reserved.



